



IECON INSTITUTO DE ECONOMÍA



The positionality of goods and the positional concern's origin

Martín Leites, Analía Rivero, Gonzalo Salas

INSTITUTO DE ECONOMÍA	
Serie Documentos de Trabajo	

Mayo, 2023 DT 04/2023

 ISSN:
 1510-9305
 (en papel)

 ISSN:
 1688-5090
 (en línea)

Esta investigación fue financiada por la Agencia Nacional de Investigación e Innovación (ANII) - subsidio FCE_3_2018_1_149516. Agradecemos especialmente a Andrea Vigorito y Guillermo Alves por sus útiles discusiones y apoyo en las etapas iniciales del proyecto. Cecilia Toledo realizó un excelente trabajo como responsable del trabajo de campo. El experimento fue registrado en el Registry for Randomized Controlled Trials operado por la American Economic Association. Identificación del RCT: AEARCTR-0003392.

Forma de citación sugerida para este documento: Leites, M., Rivero, A., y Salas, G. (2023) "The positionality of goods and the positional concern's origin". Serie Documentos de Trabajo, DT 04/2023. Instituto de Economía, Facultad de Ciencias Económicas y Administración, Universidad de la República, Uruguay.

The positionality of goods and the positional concern's origin

Martin Leites

Analía Rivero

Gonzalo Salas[‡]

Abstract

We conducted a survey-experimental on a sample of Uruguayan youth to understand what goods are positional, the degree of positional concern, and possible explanations for them. The individual's degree of positional concern was assessed by asking participants to make a series of choices between hypothetical societies characterized by varying absolute and relative income and consumption levels. We use randomized information treatments to prime participants into competing narratives regarding (i) the goods, (ii) gender, and (iii) sources of inequality in society. The main findings are: (1) the visibility of the goods would not be a necessary condition for their positionality: jewelry, cars, and health insurance are positional goods and; (2) relative income matters; (3) the positional concern is heterogeneous at the individuals level with a bimodal distribution: one group of individuals has a high prevalence of relative concern, while the other is positional-neutral; (4) there are no differences by gender in any case; and (5) individuals are less likely to report positional concerns when differences in income come from effort and inheritance.

Keywords: positional goods, visibility, meritocracy, questionnaire-experiments *JEL Code*: D63, D64, D81 C13, C91

Resumen

Realizamos una encuesta experimental en una muestra de jóvenes uruguayos para entender qué bienes son posicionales, el grado de preocupación posicional y posibles explicaciones para ello. El grado de preocupación posicional de los individuos fue evaluado mediante una serie de elecciones entre sociedades hipotéticas caracterizadas por variaciones en los niveles absolutos y relativos del ingreso y el consumo. Utilizamos tratamientos de información aleatorizados para activar en los participantes narrativas en competencia sobre (i) los bienes, (ii) el género, y (iii) las fuentes de desigualdad en la sociedad. Los principales hallazgos son: (1) la visibilidad de los bienes no sería una condición necesaria para su carácter posicional: las joyas, los automóviles y el seguro de salud son bienes posicionales; (2) el ingreso relativo importa; (3) la preocupación posicional es heterogénea a nivel individual con una distribución bimodal: un grupo de individuos tiene una alta prevalencia de preocupación relativa, mientras que el otro es neutral en cuanto a lo posicional; (4) no existen diferencias por género; y (5) los individuos son menos propensos a reportar preocupaciones posicionales cuando las diferencias de ingreso se deben al esfuerzo y la herencia.

Palabras clave: bienes posicionales, visibilidad, meritocracia, cuestionario experimental *Código JEL*: D63, D64, D81 C13, C91

[‡]M. Leites: IECON - Department of Economics - Udelar, email: mleites@iecon.ccee.edu.uy. A. Rivero: IECON - Department of Economics - Udelar, email: analia.rivero@fcea.edu.uy. G. Salas: IECON - Department of Economics - Udelar, email: gonzalo.salas@fcea.edu.uy.

1 Introduction

The idea that relative or positional concern is a key factor in understanding both individual behavior and well-being has been a long-standing issue in the social sciences. Relative concern models predict that individual preferences are interdependent, meaning that an individual's well-being and decisions are not made in isolation but rather arise from interactions with income or consumption decisions made by others and may be driven by non-pecuniary influences (Veblen, 1994; Duesenberry, 1967; Frank, 1985).¹ Extensive empirical evidence shows that relative income and consumption are important determinants of individual behavior and well-being. This evidence is consistent with several theoretical explanations. However, findings are still limited and questions remain regarding: the characteristics of goods that motivate positional concern (what types of goods are relevant to positional consumption?); the extent to which individuals care about others' consumption, and the differences in positional concern between individuals (how much do we care about other's consumption? who is more likely to make relative comparisons?); and the drivers of positional concern (why do individuals engage in relative comparisons?). This study contributes to answering these questions by testing three hypotheses suggested by previous literature within a unified and coherent framework.

Our first hypothesis postulates that the positionality of high-visibility goods' is higher than that of lowvisibility goods, and both are lower than the relative degree of income concern. This hypothesis is grounded in the literature on the attributes of positional goods – items with values that depend on how they compare with items owned by others (Frank, 1985; Hirsch, 1976). Theoretical models and previous evidence support the idea of desirability and socio-cultural visibility (for a review Heffetz and Frank (2011)). There is less evidence to explain why, in certain contexts, the consumption of one good is positional while others are not. Our second hypothesis states that the degree of positionality is an individual attribute. In this case, we assess how persistent an individual's positional parameter is when he/she presents a high relative income concern. This hypothesis is inspired by literature that suggests that positional social arrangements induce individuals to compete and make relative comparisons (Hopkins, 2008). In this case, the formation of preferences and the adoption of cultural norms lead to people presenting heterogeneity in the degree of positionality (Bisin and Verdier, 2011). The third hypothesis focuses on the drivers of individual positional concern. On the one hand, based on evolutionary approaches, we posit that men exhibit greater positional concern than women. On the other hand, we investigate whether the origins of inequality are a significant driver of the variability in the degree of positional parameters between individuals. This line of questioning is motivated by the idea that relative concerns can be strongly influenced by a desire for fairness and an aversion to inequality (Clark and D'Ambrosio, 2015; Fehr and Schmidt, 2003).

Testing for the importance of relative income/consumption for individual well-being, as well as why (and when) individuals are willing to sacrifice absolute income/consumption for a higher relative position, is

¹Some models assume that individuals value relative standing because it is a direct argument in their utility function. In this case, relative standing directly affects individuals' well-being. Other models assume that relative standing could have an instrumental value. For instance, consuming certain goods could be seen as a sign of high status that the individuals then use to obtain other advantages. Both models provide foundations to explain relevant externalities yielded by the interdependence of individuals' decisions with well-established welfare consequences (Frank et al., 2005; Frank, 1985). Complete reviews of more recent literature on the well-being implications of relative or positional concern can be found in Weiss and Fershtman (1998); Postlewaite (1998); Heffetz and Frank (2011).

a challenging endeavor for several reasons. On the one hand, measuring the positional parameters with direct questionnaires and understanding their main drivers requires making restrictive assumptions. On the other hand, since alternative foundations have been proposed in the literature to explain relative or positional concerns, it is difficult to identify the competing mechanisms. In order to address these issues, we employed survey-experimental methods on a sample of over 900 young people in Montevideo to elicit their degree of positional concern with respect to income and consumption. Participants were recruited through the Longitudinal Welfare Study in Uruguay (ELBU) and come from different socioeconomic levels. Beshears et al. (2008) and Amiel et al. (2015) argue that experimental surveys are useful strategies to measure agents' norms, values, and goals.

Our questionnaire-experimental design was adapted from Alpizar et al. (2005) and Amiel and Cowell (1992). We elicited the degree of individual positional concern by asking participants to choose between hypothetical societies characterized by varying absolute and relative income/consumption levels. Participants were asked to choose among five pairs of societies in which they would prefer their grandchild to live. The different pairs of societies are characterized by different average levels of consumption and grandchild consumption expenditure levels, allowing for a trade-off between an individual's absolute and relative consumption expenditure. We manipulated the information about hypothetical scenarios considering three goods with different characteristics: cars (high visibility and high use value), jewelry (high visibility and low use value), and health insurance (low visibility and high use value). As a benchmark, we also used the degree of positionality of income (low visibility and determinant of consumption capacity). This allowed us to provide new evidence on which goods are positional and compare their degree of positionality with the relative income concern parameters. If relative income concern is partly driven by inequality aversion, the difference between the parameters of the positionality of the goods and the income provides a lower bound of the positional concern associated with the consumption of goods.

Additionally, we explore the heterogeneity of positional concern by gender and within individuals for different types of goods. To advance our understanding of the drivers of a positional concern, we introduce two information treatments that have not yet been explored in the context of the relative income/relative consumption experimental survey. First, to test for gender differences in the positionality parameter, a random sample of respondents was asked about their granddaughter's or grandson's situation. Then, following Bergolo et al. (2022), we carried out the relative income experiment with an additional information treatment for the origin of the grandchild's income: effort or inheritance. Both treatments - the *effort-message* and the *inheritance-message* - allow us to test the sensitivity of the positionality parameter to individual beliefs about the role of meritocracy. As a result, this treatment allows us to assess whether the respondents' marginal degree of income positionality is elastic to the notion of fairness.

Our survey collects data about socioeconomic and demographic characteristics and includes questions regarding respondents' social backgrounds, opinions, attitudes, and preferences. We also include a series of questions used in economics to measure interpersonal comparison and compare the individual positional parameter elicited by the experimental survey with the individual alternative measures based on direct questions. This allows us to assess the consistency between both measures.

Regarding the first research question, our results show that despite their differences, jewelry, cars, and

health insurance are highly positional goods. The marginal degree of positionality for these goods for the median respondent belongs to the interval [0.75, 1), and depending on the specification, the estimates of the mean are around 0.82 - 0.90, 0.85 - 0.92, and 0.76 - 0.91 respectively. In all estimated models, the average marginal degree of the positionality of the goods –cars, jewelry, and health insurance– is between 19% and 36% higher than the positionality of income (the differences are statistically significant at 1%). The evidence for our second hypothesis is in line with previous findings. Our relative income experiment survey confirms that most individuals care about their relative income: the marginal degree of positionality for income for the median participant belongs to the interval [0.5, 0.75), and the estimate of the mean is around 0.6. This means that, on average, an increase in personal income generates more utility due to the increase in their relative income than an increase in the absolute income (around 50%). Additionally, our results suggest that the degree of positionality is an individual attribute and shows a strong heterogeneity of positional concern at the individual level. The distribution of the marginal degree of positionality is bimodal for the selected goods and income, with one group having a high prevalence of relative concern in all goods and the other being positional-neutral. Finally, we explore the drivers of positionality concerns to address the third hypothesis. We reject the presence of significant differences by gender. Additionally, our study reveals that individuals are less likely to report positional concerns when inequality/position comes from effort and inheritance. This result suggests that positional parameters incorporate inequality aversion and are sensitive to individuals' sense of fairness.

This study contributes to the literature on positional concerns. There is considerable evidence about individual relative concern from the different corpus of literature as empirical studies of expenditure (Agarwal et al., 2021; De Giorgi et al., 2020; Chai et al., 2019; Bertrand and Morse, 2016; Kaus, 2013; Charles et al., 2009), natural and field experiments (Dannenberg et al., 2022; Cullen and Perez-Truglia, 2022; Perez-Truglia, 2020; Bursztyn et al., 2018; Kuhn et al., 2011), happiness (Perez-Truglia, 2020; Clark and D'Ambrosio, 2015; Clark et al., 2008; Ferrer-i Carbonell, 2005), lab experiments (Fehr and Schmidt, 2003), and experimental surveys in which participants choose between hypothetical situations (Carlsson et al., 2010, 2009, 2007; Alpizar et al., 2005). This paper directly relates to this latter group of studies. Alpizar et al. (2005) investigate the relevance of an individual's absolute and relative position in income and consumption of particular goods. They find that cars are more positional than leisure and insurance. Additional studies by Carlsson and other coauthors on other countries and samples conclude that the degree of positional concern varies between societies (Carlsson et al., 2010, 2009, 2007).

We use a unified framework to accurately measure positional concerns in different dimensions among a diverse sample of young people and assess whether individuals' positional concern is consistent for goods with different characteristics. We confirm that positionality is an individual attribute, and a high relative income concern is associated with high positionality in consuming a set of goods. This paper offers new insights into the drivers of positional concern. Our results suggest that positional concerns may be motivated by ideas of fairness. Finally, the high degree of positionality found for these goods holds when we discount the maximum potential effect of income inequality aversion.

Indirectly, these results relate to the literature that has explored the drivers of inequality aversion (Bergolo et al., 2022; Almås et al., 2020; Cappelen et al., 2020; Carlsson et al., 2005). Previous research has examined how people's attitudes toward inequality change when they acquire information regarding the source

of income inequality. However, the degree to which inequality aversion is sensitive to the inheritance of income is scarcely studied. Using randomized experiments, Bastani and Waldenström (2021) and Fisman et al. (2020) (for Sweden and for the United States, respectively), found that individuals prefer a higher tax rate for wealth when the source of wealth is inheritance. We build on this literature by using an alternative strategy that combines the experimental-questionnaire approach with information treatments in order to test whether inequality aversion responds differentially to inheritance and effort. Our findings suggest that positional concern, driven by inequality aversion, is significantly lower when inequality is perceived to result from effort and inheritance. A novel result is that inheritance tends to be perceived as merit. Consequently, we find that participants are more tolerant of inequality is unknown and they decide based on their prior beliefs. The origin of this difference could be related to the fact that this study measures aversion to inequality rather than preferences for tax rates. Furthermore, in the context of our experimental questionnaire, a grandchild's inheritance could be perceived as the result of the effort of the participant. This evidence suggests why there is no support for imposing high inheritance tax rates, which generally have marginal weight.

Our study is also related to the literature on the attributes of positional goods. Evidence provided by Heffetz (2018, 2011) supports the idea that socio-cultural visibility is a relevant characteristic of positional goods. It is not driven by 'objective' observability but depends on the cultural and social value each society or socioeconomic group assigns to holding certain goods and the associated expenditure. Most papers use visibility surveys that focus on the noticeability of the average spending of a list of goods (Heffetz, 2018; Kaus, 2013; Khamis et al., 2012; Heffetz, 2011; Charles et al., 2009; Chao and Schor, 1998). Heffetz (2018) expands the notion of visibility and concludes that spending on some goods generates different positive/negative impressions conditionally on being noticed. Alves et al. (2022), use data collected in a field experiment and a survey sample of youngsters and suggest that not all socially valued goods present high social visibility.

We confirm a high positionality for high visibility and luxury goods, in line with the previous findings by Heffetz (2018, 2012); Charles et al. (2009); Alpizar et al. (2005). Unlike Alpizar et al. (2005), a high positionality is confirmed for a low visibility-good like health insurance – the magnitude of the positional parameter is two times higher in our study. As for Alves et al. (2022), our results suggest that positional concern is not directly associated with the social visibility of expenditures.

The contribution of this study is related to the previous literature that explores differences in psychological attributes and traits by gender (Bertrand, 2011). Contrary to what some evolutionary approaches suggest, the various tests carried out reject the existence of differences in degrees of positionality based on gender.

Finally, this paper provides some methodological contributions to studies that make use of experimental questionnaires to elicit behavioral parameters (Carlsson et al., 2010, 2009, 2007; Alpizar et al., 2005). First, this paper suggests that these instruments are useful for a sample of non-university populations. The results are robust for the number of desired children of the respondents and the use of different hypothetical scenarios, such as the sex of the grandchild. To assess the validity of a positionality measure

elicited from an experimental survey, we analyze whether it is correlated with a wide range of selfreported variables related to preferences for status. Additionally, the study takes into consideration the potential consequences of inconsistent responses when determining positionality parameters.

The rest of the paper is organized as follows. Section 2 summarizes the foundation of relative concern and the mechanisms that could explain its origin. Section 3 explains the main details of our experimental design. Section 4 describes the experiment implementation and the information collected. Section 5 reports the main results. In section 6, the validity of our results is discussed. Section 7 concludes.

2 Foundations of positional concern: an organizing framework

The notion that individual decisions are often driven by positional concern is not a new concept (Veblen, 1994; Frank, 1985; Duesenberry, 1967). People derive satisfaction not only from their absolute income but also their relative income position in society or a particular social group. Consumption decisions are not made in isolation from the consumption choices of others. For certain goods, an individual's utility depends not only on their own consumption level but also on the consumption levels of others. Positional concern reflects how individuals care about the consumption or income of others.

The existence of positional concerns is consistent with several theoretical explanations. The possible reasons why people have positional concerns are also diverse. Evolutionary theory provides a strong argument for an innate perspective (Frank et al., 2005; Postlewaite, 1998). It is based on the idea that a higher position in the past increases the possibility of success (and survival) compared to individuals in lower positions. In this case, the relative concern is intrinsic to human behavior motivated by rivalry and interpersonal competence. Hopkins (2008) suggests two other reasons. The concept of status can be viewed as a visible signal sent to society to improve well-being or gain some advantage. This is a consequence of incomplete information or preferences, as the consumption or performance of other individuals becomes relevant for making better decisions. Another explanation is that relative concern arises from current social arrangements, where the nature of institutions and social norms induce individuals to compete and make relative comparisons. Related to the latter explanation, Bisin and Verdier (2011) points out that individuals develop positional preferences through observation, learning, imitation, and by adopting cultural norms. Relative income/consumption concern could be strongly motivated by concerns for fairness and inequality aversion due to fairness motivations (Clark and D'Ambrosio, 2015; Fehr and Schmidt, 2003). In this case, the positional concern is explained by inequality aversion and unfair distribution.

Finally, instrumental motivations can affect relative income/consumption. For instance, consuming certain goods can signal individuals' social status. The relative consumption of these goods might serve as a sign of economic success and could consequently have an instrumental value in power, preferential treatment (or discrimination), or permit the individual to gain access to jobs or prestigious positions. The positional concern could also be explained by imperfect information about prices, wages, and the quality of the goods, particularly as knowledge of others' consumption patterns could provide relevant information about economic status. Another potential source of relative concern emerges when individual preferences are incomplete. The own experience and the experience of others is a fundamental mechanism to evaluate objects, opportunities, or income (Hopkins, 2008). In this case, through relative comparisons, people shape their preferences. Finally, the concern about the consumption of others could be motivated by the network effect (Heffetz and Frank, 2011).

Information about others' income (consumption) is key to understanding an individual's utility and choices. Since relative concerns imply that an individual's utility could be affected by the income or consumption of others, they could produce externalities.

This interdependence of individual decisions generates externalities with micro and macroeconomic implications. On the one hand, when an individual decides to increase their consumption because they are motivated by positional status, negative externalities are produced for those surrounding them. On the other hand, the competition for status leads to the over-consumption of positional goods (Heffetz and Frank, 2011).

A general formulation of the utility function U of an individual *i* that incorporates relative concern is as follows:

$$U_{i} = v_{i}(x_{i}^{b}, R_{i}(x_{i}^{b}, x_{-i}^{b}))$$
(1)

where x_i^b and $R_i(x_i^b, x_{-i}^b)$ represent the absolute and relative endowment of *b* (income or certain good). Previous literature uses different functional forms of R_i to model positional concerns. As we explain in more detail in the section 3, we assume a mean-dependence model. Namely, the relative component R_i assumes that individuals compare their income/consumption with the average income of others (van Praag, 2011; Clark and Oswald, 1998, 1996; Abel, 1990; Boskin and Sheshinski, 1978; Duesenberry, 1967).² The mean-dependence model incorporates a notion of self-centered inequality aversion because it assumes that individuals care about the differences between their income and that of others (Burone and Leites, 2021; Alpizar et al., 2005; Johansson-Stenman et al., 2002). In this case, the willingness to pay to reduce inequality is centered on the individual's situation in relation to the mean.³

To measure positional concern, we follow Alpizar et al. (2005) and use the marginal degree of positionality as:

$$\gamma_i^b = \frac{\frac{\partial v_i}{\partial R_i} \frac{\partial R_i}{\partial x_i^b}}{\frac{\partial v_i}{\partial x_i^b} + \frac{\partial v_i}{\partial R_i} \frac{\partial R_i}{\partial x_i^b}}$$
(2)

This ratio compares the change in the utility of a marginal increase in relative income/consumption with the total change in the utility of a marginal increase in income/consumption. Two elements are key

²Other authors include relative income concern based on rank (Clark et al., 2009a,b; Robson, 1992; Frank, 1985; Layard, 1980).

³Models based on inequality aversion assumes diverse functional form for R_i , for example, including the global inequality (coefficient of variation or Gini index) or the relative affluence or deprivation. Febr and Schmidt (2003) and Clark and D'Ambrosio (2015) provides a detailed review about inequality aversion models. The inequity aversion model of Febr and Schmidt (1999) is canonical. Hopkins (2008) demonstrates that mean dependent models are a special case of the social preference model proposed by Febr and Schmidt (1999).

in equation 2. The first one is that individuals' marginal degree of positionality varies between goods. Hirsch (1976) and Frank (1985) define positional goods as "those things whose value depends relatively strongly on how they compare with things owned by others." We identify x_i^b as a non positional good when $\gamma_i^b = 0$, while x_i^b is a positional good when $\gamma_i^b > 0$.

As mentioned above, our measure of γ_i^b might incorporate a notion of non-self-centered inequality aversion. As a benchmark, we take the degree of positionality of income, γ_i^{income} . We assume that it provides an upper bound for the role of inequality aversion. The gap between γ^b and γ_i^{income} provides a lower bound of positional concern motivated by positional consumption. We identify x_i^b as a consumption positional good when the difference $\gamma_i^b - \gamma_i^{\text{income}}$ is statistically significant and positive.

Second, as with other social preferences, individuals could have different degrees of positional concern between them. γ_i^b is the parameter reflecting how people care about the utility of others. Under this specification, γ may vary between individuals in the same society and for the same endowment x_i^b . Namely, if $\gamma_i^b > \gamma_i^b$, the individual *i* has more positional concern than *j*.

Another major point is about the benchmark or the relevant reference group that individuals use to make interpersonal comparisons. It is worth mentioning that an individual may have more than one reference group. Even which goods an individual considers positional may vary according to the relevant reference group in a given context.

As a result, the identification of the relevant reference group is an unobserved variable. As described below in section 3, the experimental survey applied in this paper addresses this empirical problem by assuming a single reference group that does not vary between individuals and different goods. This assumption allows the results of the relative concern to be comparable. However, the experimental design does not reveal which group is relevant for each individual. This limitation should be taken into account when interpreting the results.

3 Experimental design

Testing whether people have positional concerns and why they choose to reduce their absolute income or consumption for more relative income or consumption is not a simple task. Amiel et al. (2015) and Beshears et al. (2008) argue that experimental surveys provide useful information about agents' preferences, norms, values, and goals. To estimate the magnitude of the marginal degree of positionality at the individual level, we implemented an experimental survey with a sample of 951 youngsters aged 18-20 from Uruguay. Unlike most previous studies that relied on a sample of university students, we interviewed the participants face-to-face at their homes during this fieldwork. This difference might reduce the possibility of social desirability bias (Holbrook et al., 2003).

The baseline experimental survey consisted of three sections: (i) the relative consumption experiment; (ii) the relative income experiment; and (iii) a final module that collects information on respondent's socioeconomic status, attitudes, beliefs, goods visibility scales, and self-reported preferences for social status and reference groups.⁴ In sections (i) and (ii), the participants were instructed to consider the

⁴The entire Spanish version of the questionnaire is available in https://fcea.udelar.edu.uy/

well-being of their imagined grandchild when making choices, following the approach of Alpizar et al. (2005) and Johansson-Stenman et al. (2002). The participants were presented with pairs of hypothetical societies and had to make five decisions. In section (i), entitled "relative consumption experiment," the pairs of societies were characterized by the consumption of a particular good; while in section (ii), the "relative income experiment," the societies were described by income. In all cases, the participants were given the same baseline instructions and were told which level of income/consumption their grandchild would have and the mean of relative income/consumption in society.

In addition, the study used randomized information treatments to prime participants into competing narratives regarding (a) the goods; (b) sources of inequality in society; and (c) grandchild gender. The goods treatment was carried out in the relative consumption experiment to understand the heterogeneity of positionality between goods. In the relative income experiment, additional information regarding the sources of inequality was introduced (effort or inheritance). Finally, the third treatment was based on the idea that the responses could be sensitive to the gender of the grandchild.

Following previous empirical social choice research, participants made hypothetical choices without real financial incentives. This may raise concerns about the reliability of our questionnaire-based measure of positional concern. Individuals may engage in cheap talk or provide socially desirable answers. They may look more prosocial compared to situations wherein their decisions have a personal cost. However, according to Gaertner and Schokkaert (2012), strategies based on experimental questionnaires provide useful information about agents' norms, values, attitudes, ethical preferences, and goals. Moreover, individuals would provide reliable answers, especially if the tasks performed by participants are not cognitively costless. There could be "mistakes" in the choices by some participants if the questions are particularly difficult to understand; however, it is expected that the majority of participants should try to answer the questions carefully and truthfully if the questionnaires are sufficiently clear and respondents volunteer to participate (Bauer et al., 2020; Larney et al., 2019; Ben-Ner et al., 2008).

3.1 Eliciting positional concern

To elicitate the marginal degree of positionality, we use the ratio comparison utility function in Alpizar et al. (2005). In the basic model, individual *i* derives utility from their own income/consumption and the ratio between the income/consumption and the average income/consumption of the society in which they live. Note that this strategy assumes that there is a unique and common comparison benchmark. The general formulation of this for an individual *i* who lives in a society *s* is:

$$v_{i,s} = \left[x_{i,s}^b\right]^{(1-\gamma_i^b)} \cdot \left[\frac{x_{i,s}^b}{\bar{x}_s^b}\right]^{\gamma_i^b}$$
(3)

where $x_{i,s}^b$ is the level of income/consumption of good *b* corresponding to individual *i* living in society *s*, \bar{x}_s is the average income/consumption for society *s* and γ_i^b is a parameter of the individual degree of positionality regarding the income or the good *b*. Under this specification, γ_i can be interpreted as

investigacion/proyectos/estudio-longitudinal-de-bienestar-en-uruguay.html.

a marginal degree of positionality, namely the proportion of the total utility variation yields from an increase in the relative consumption (or income) from the last unit of money spent (obtained). In the extreme case where $\gamma_i = 0$, individual *i* does not care about relative income or consumption. When $\gamma < 0$, individual *i* obtains utility from relative deprivation, i.e., compassion effect. When $\gamma_i > 0$, individual *i* is concerned with both the absolute and relative consumption (income) of particular goods. This means that even for a marginal increase in income or consumption, there is an increase in utility due to the absolute and relative effects. For example, if we assume that $\gamma = 0.4$, an increase in individual *i*'s consumption generates an increase in their utility, with 60% of this increase being explained by the absolute component and the remaining 40% being explained by the relative consumption (income) than for absolute consumption (income).

Consider a hypothetical choice between two societies, A and C. Both societies are identical –including the level of inequality, the availability of goods and services, and the same prices and quality– except for their average income/consumption and the grandchild's income/consumption. If we assume the individual utility function defined equation 3, only when the following equality holds an individual is indifferent between these two societies:

$$v_{i,A} = \left[x_{i,A}^{b}\right]^{(1-\gamma_{i}^{b})} \left[\frac{x_{i,A}^{b}}{\bar{x}_{A}^{b}}\right]^{\gamma_{i}^{b}} = \left[x_{i,C}^{b}\right]^{(1-\gamma_{i}^{b})} \left[\frac{x_{i,C}^{b}}{\bar{x}_{C}^{b}}\right]^{\gamma_{i}^{b}} = v_{i,C}$$
(4)

From the indifference condition in equation (4), we know that:

$$\hat{\gamma}_i^b = \frac{\log(x_{i,C}^b/x_{i,A}^b)}{\log(\bar{x}_C^b/\bar{x}_A^b)}$$
(5)

This equation establishes a clear trade-off between individuals' income/consumption and average income/consumption in a society. This means that for *i* to be indifferent between societies *A* and *C*, a decrease in relative position may be compensated by some additional absolute income/consumption, such that the overall level of utility remains constant. This equation allows us to recover the marginal degree of positionality parameter from individual choices. For example, consider a society *A* where the individual's income is \$37,500 per month and the average income is \$45,000, while in society *C* they are \$27,500 and \$30,000 respectively. Holding other things equal, the implied value of γ associated with indifference between the two societies *A* and *C* is $\hat{\gamma}_i^b = \frac{\log(27,500/37,500)}{\log(30,000/45,000)} = 0.75$. This implies that when an individual prefers *C* over *A*, we know that $\gamma \ge 0.75$, while when she prefers *A*, $\gamma \le 0.75$.

The participants made five consecutive choices between a fixed society A and alternative societies C_z . It is worth noting that eliciting the participants' positionality parameters requires their choices (at least) to hold weakly monotonic preferences. Namely once they choose A over C_z , they should not choose C_{z+1} over A again. Furthermore, some participants responded consistently to some of the experimental surveys but were inconsistent in the rest. This generates an additional restriction. We analyze this point in section 4.2. If we assume that the societies A and C_z are identical in everything, including their levels of inequality, the elicited positionality parameter does not consider non-self-centered inequality aversion. The differences in the relative gap regarding the average individual could motivate aversion to self-centered inequality. This agrees with the positional concern and, as mentioned above, will be considered in interpreting the parameters.

3.2 Experimental design

In the first stage of the experiment, the interviewer reads the instructions, explaining each part. Participants face repeated pairwise choices between hypothetical societies in this experimental survey. They receive information about both societies in a baseline instruction.⁵ We define a baseline society A and five alternative societies C_z . The hypothetical societies are characterized by two dimensions that correspond to the arguments of the utility function in equation (3): income/consumption of good b (x_i^b) and average income/consumption of good b in society s (\bar{x}_s^b). In all other respects, the instructions explicitly mention that the alternatives are identical in all other respects.

Following the approach of Johansson-Stenman et al. (2002) and Carlsson et al. (2005), the participants were instructed to choose a society that would result in better well-being for their imagined grandchild eighty years from the time of the experiment. This practice is common in survey experiments and aims to eliminate the influence of the respondents' current circumstances or social environment on their decisions. As most of the participants were between 18 and 20 years old at the time of the survey, they were not old enough to have real grandchildren. The assumption underlying this approach is that participants will base their choices on their own preferences or anticipate their imagined grandchild's preferences to be similar to their own (Johansson-Stenman et al., 2002).

The interviewers read the instructions before the first pair-wise choice between A and C_1 . Then interviewers show a card with a description of both societies in a comparable way, and they ask participants to choose between both hypothetical societies. The information for each society includes the mean and the grandchild's income/consumption.

For example, if we focus on the relative income experiment, society A is characterized by a grandchild's income of \$35,000 and an average income in the society of \$45,000. Each type of C_z society has an average income of \$30,000, representing the 67% of the average income reported for society A. The only difference among type C_z societies is the income the grandchild would receive. By changing the grandchild's income and holding the relative income constant, we can elicit bounds for each respondent's marginal degrees of positionality parameter.

Following the example of the relative income experiment, Table A1 in the Appendix describes each of the societies regarding the grandchild's income and the average income (columns 2 and 3). By asking individuals to make repeated pair-wise choices between hypothetical societies with different implicit marginal degrees of positionality (column 4), it is possible to elicit that parameter at the individual level.

⁵The card shown to the participants in the case of the relative income experiment is shown in Figure A1 of the Appendix. The instructions explicitly mention that there are no right or wrong answers. We also told participants that everyone in these societies could cover their basic needs so as to rule out poverty aversion or lexicographic strategies.

The following example illustrates how we identify the lower and upper bounds for γ in the case of the relative income experiment. Let a set of choices be, for instance, $\{C_1, C_2, A, A, A\}$. This implies that $C_2 \geq A$ and $A \geq C_z, \forall z > 2$. From the indifference condition in equation (4), we know that $\gamma \geq 0.125$ and $\gamma \leq 0$. The intervals for γ associated with each possible (and consistent) set of choices are reported in Table A1. It is worth noting that if subject *i* chooses *A* over C_1 , she is choosing to resign part of her income to increase her relative income. We call this type of subject "positional individuals." In any other case, individuals can be defined as positional-neutral.

3.3 Information treatment

The baseline instruction, which is common in all cases, establishes:

Baseline-message: "Next, imagine that 80 years have passed and you have the chance to choose in which society your only grandson/granddaughter will live. Assume that all individuals have their basic needs met in all the societies listed below. There are no wrong or right answers. We ask you to carefully consider your preferred alternative for your grand-child and not the best society in general. You can select between pairs of societies that are identical in all, except..."

Relative income experiment. The respondents choose among several pairs of hypothetical societies, A and C_z , described by the average income and the grandchild's income.

Baseline-message: "You can select between pairs of societies that are identical in every way, except in your grandson/granddaughter's income and the average income of the other people in the society"

Relative consumption experiment: treatment goods. To analyze the role of alternative goods on the positional concern, we carried out the experiment of relative consumption. Like Alpizar et al. (2005), the participants choose between two societies that describe the grandchild's consumption and the average consumption of a particular good. In all other aspects, both societies were identical. We use the responses of the experimental-questionnaire approach and strategy mentioned in Section 3.1, where we elicit γ^b for each good *b*.

By maintaining comparable societies in terms of absolute consumption and relative consumption for a set of goods, the structure of the choices allows us to identify the implied γ^b in the same range of values. This range is also comparable with the relative income experiment.

Our treatment of information is designed to identify if an individual's positional concern is affected by a change in the relative position dimensions –the goods– keeping constant the rest of the conditions. The structure of the experimental questionnaire is identical for all participants. The treatment of information about the change of good is the unique exogenous variation. Since participants are exposed to the same sequence of decisions for a set of goods, we can compare the effect of the good on an individual's marginal degree of positionality at the individual level. This allows us to elicit the individual's marginal degree of positionality for the consumption of four alternative goods and analyze whether there are differences in

the degree of positionality among these goods.

A key step is selecting the good used to measure the degree of positionality. The positional goods must meet at least three essential characteristics: i) the access and consumption of these goods are associated with social recognition, the esteem of others, and success in life. ii) the social visibility of consumption of these goods, a characteristic suggested by the theories of signaling and conspicuous consumption; iii) by definition, the relevance of relative consumption of that good makes no sense in social isolation. It is an inherently spatial-temporal phenomenon (in a social sense) that depends on social interactions and relevant reference groups.

Another criterion for selecting goods was that they allowed some comparability with previous studies using similar strategies. Alpizar et al. (2005), for example, use cars, housing, insurance, and vacation days. Additionally, we incorporate some changes considering the interests and consumer profile of young people in Uruguay. For most young people of this age, it is not expected that they would be interested in buying a house, while it is more plausible that they would be interested in buying a car. For this reason, we consider the second good. The selection of goods considers the visibility index developed by Heffetz (2011), which was applied to the same sample of youngsters in 2015/16 (Leites et al., 2019). We also conducted a series of focus groups before the survey with a similar population of youngsters. Jewelry appeared in the focus groups as a socially visible good associated with a high socioeconomic position and is the most visible good in Heffetz's list. We also include health insurance, which has lower social visibility. Since access to health services is guaranteed for all people in the Uruguayan health system (both with private and public providers), individual health insurance represents extra coverage for specific services or an extension of some rights (for example, different hotel services when people are hospitalized). Access to individual health insurance implies an additional price and is generally associated with those workers with better jobs or families with higher incomes.

In sum, we include three different types of consumption in the experiment: cars (high visibility and high use value), jewelry (high visibility and low use value), and health insurance (low visibility and high use value). The instruction said:

Goods-message: "You can select between pairs of identical societies, except on how much the average spend on [health insurance premiums/Jewelry/cars] and how much your grandson/granddaughter spends for this purpose. Although other people spend more (on average) on [health insurance premiums/jewelry/cars] in society A, their consumption of other goods is the same in both societies. Please choose in which of the societies I show you below your [grandson/granddaughter] will be better off"

Fairness treatment: effort and inheritance. Following Bergolo et al. (2022), we carried out the relative income experiment, but we included additional information treatment regarding the sources of inequality. The two treatments - *effort-message* and *inheritance-message* - are based on the idea that relative concern and inequality aversion are sensitive to a notion of fairness. This message is shown to the participants immediately after they complete the baseline relative income experiment. The interviewers read the new instructions, and the participants make the sequences of choices between A and C_z with additional information about the source of inequality. The first group of respondents chooses between hypothetical societies, where their choices determine their grandchild's income and relative income. In this case, the instruction includes an information treatment that reveals that the origin of the grandchild's income is their effort. The second group makes the same choices, but the instruction includes an information treatment establishing that their grandchild's income from inheritance. The *effort* and *inheritance* messages are as follows:

Effort-message: "You can select between pairs of societies that are identical in all, except in your grandson/granddaughter's income and the average income of the other people in the society. In addition, you know that your grandchild's income and his/her place in society correspond to his/her lifelong effort relative to the others"

Inheritance-message: "You can select between pairs of societies that are identical in all, except in your grandson/granddaughter's income and the average income of the other people in the society. In addition, you know that the inheritance mainly explains your grandchild's income he/she received "

Grandchildren's gender information treatment. To test for gender differences in positionality parameters, a random sample of the respondents is asked about their granddaughter's or grandson's situation. We randomly divide the sample into three groups. A group of participants in our baseline experimental survey does not distinguish the sex of the grandchild. In all cases, the text said grand-son/granddaughter. The second group of respondents chooses between hypothetical societies, where their choices determine their grandson's income and relative income. The third group of respondents chooses between hypothetical societies, where their choices determine their grandson's income and relative income. In sum, the text said:

Control: "... the chance to choose in which society your only grandson/granddaughter will live"

Grandson: "... the chance to choose in which society your only grandson will live"

Granddaughter: " ... the chance to choose in which society your only granddaughter will live"

In both cases, we adapt the text of the rest of the instruction to the grandchild's gender. As we explain 4 we applied these two alternative questionnaires to a sub-sample.

3.4 Hypotheses

Regarding the positionality of the goods, our leading hypotheses for the treatment arms presented in the previous section are:

$$0 < \gamma^{\text{income}} \le \gamma^{\text{visible goods}};$$
 with visible goods = {Cars, Jewelry} (H1-a)

where γ^{income} , $\gamma^{\text{insurance}}$ and $\gamma^{\text{visible goods}}$ represent the value of the positional parameter estimated for the relative income experiment, health insurance experiment, car experiment and, jewelry experiment.

As we discussed above, in our context, the magnitude of the parameter γ^{income} includes the influence of self-centered inequality aversion. Furthermore, although the relative income experiment refers to the capacity to consume, it does not mean making an expenditure. As an additional test to identify the positional concern associated with the consumption, we test a complementary hypothesis (H1-b):

$$\gamma^{\text{goods}} - \gamma^{\text{income}} > 0$$
; with goods = {Cars, Jewelry, Insurance} (H1-b)

We explore whether the degree of positionality is an individual attribute. To advance in this direction, we assess how persistent individuals' positional parameters are when we consider the relative consumption experiments concerning the relative income experiment. For example, there could be participants for whom the level of income positional concern is low (l) and is not sensitive to the type of goods. Or, individuals could exhibit a high level of positional concern (h), irrespective of whether we consider income/goods. Finally, there could be participants for whom income positional concern is low but high relative consumption concerns. This establishes the hypothesis that those participants with a high degree of income positionality present greater positionality in regard to cars, health insurance, and jewelry. In more formal terms, this implies:

$$F_i(\gamma^{\text{goods}} \mid \gamma_h^{\text{income}}) \ge F_j(\gamma^{\text{goods}} \mid \gamma_l^{\text{income}}) \iff \gamma_h^{\text{income}} > \gamma_l^{\text{income}}; \text{ with goods} = \{\text{Cars, Jewelry, Insurance}\}$$
(H2)

where F represents the distribution γ^{goods} of the respondents (*i* and *j*) conditional to the values of γ^{income} .

Finally, we explore some drivers of the positional parameter. First, we explore the hypothesis that the degree of positionality varies by own and grandchild's gender. Previous papers document a gender gap in performance when individuals are exposed to alternative competitive environments and consider social status ranking. Differences in the positionality parameter could explain these differences in behavior. We expect that the marginal degree of positionality is lower for females than males and granddaughters than grandsons.

$$\gamma_f^b \le \gamma_m^b$$
; with b = {Income, Cars, Jewelry, Insurance} (H3-a)

where γ_f^b and γ_m^b represent the positional parameter for females and males, respectively.

Second, we explore whether the sources of inequality are a relevant driver of the heterogeneity in the degree of positionality parameter across individuals. This analysis is motivated by the idea that the individual's relative income concern is directly related to his/her self-centered inequality aversion. If this is true, we expect that the marginal degree of positionality will be sensitive to the individuals' notion of fairness and their beliefs about the source of inequality. We randomly introduce information about

the differences in the source of relative income to test how positional parameters and fairness relate to each other. The baseline instructions (without additional information) represent the control group. Two additional groups randomly received the two fairness informational treatment: effort and inheritance. The details of this strategy are described in subsection 3.3 and 3.5. Following Bergolo et al. (2022), treatments aim to test the idea that individuals are more likely to accept inequality from the differential effort, while they are more reluctant when it comes from circumstances beyond individual control as an inheritance.

Our leading hypothesis for the *effort-message* arms is:

$$\gamma^{\text{income-effort}} - \gamma^{\text{income}} < 0 \tag{H3-b}$$

where $\gamma^{\text{income-effort}}$ and γ^{income} represent the value of the positional parameter estimated for the *effort-message*, and *baseline instructions* respectively.

The *inheritance-message* has two possible opposite effects. On the one hand, inheritance could be perceived by the respondents as a circumstance beyond the individual's control that is obtained without any merit of the heir. On the other hand, inheritance could be perceived as a fair intergenerational transmission explained by parental' effort. While in the first case, the effect of *inheritance-message* implies an increase in the positionality parameter, in the second, it is expected a given these opposite effects, the expected sign of this message is undetermined:

$$\gamma^{\text{income-inheritance}} - \gamma^{\text{income}} = ?$$
 (H3-c)

where $\gamma^{\text{income-inheritance}}$ represents the value of the positional parameter estimated for the *inheritance-message* group. Beyond the sign of these hypotheses, the confirmation that the magnitude of the positional parameter is sensitive to a notion of fairness suggests the relevance of self-centered inequality aversion. This reinforces the relevance of testing H1-b to identify the magnitude of the positional effect associated with consumption.

3.5 Econometric specification

This section presents the econometric strategy used to test our hypotheses. In the first step, the aim is to identify what influences the degree of marginal positionality using the information treatment about the goods described in the previous section. Since all participants make the same choices, we compare their choices when considering income and the different types of goods.

As preliminary step, we use a statistical test to assess whether the mean positional parameters by goods are statistically significantly different from zero. This provides a direct test for hypothesis H1-a.

Hypothesis H1-b incorporates as a benchmark the marginal degree of the positionality of income. To test that hypothesis, the main specification is given by the following regression:

$$\vec{\gamma}_i^b = \Gamma(\alpha + \beta^{\text{good}} \cdot D_i^{\text{good}} + \mu_i + \varepsilon_i^b) \tag{6}$$

The outcome variable (γ_i^b) represents the parameter of the marginal degree of positionality recovered from the set of choices of societies *A* and C_z made by the participants in the several alternatives. For each participant *i* the vector $\vec{\gamma}_i^b$ contains the γ_i^{income} , $\gamma_i^{\text{jewelry}}$, $\gamma_i^{\text{insurance}}$ and γ_i^{car} . The former was elicited based on equation (5) and relative income experiment responses. The other three parameters were elicited based on equation (5) and the relative consumption experiment. D_i^{good} is a dummy variable indicating whether participant *i* completes the experiment for the visible and nonvisible good. Finally, Γ is a generic function that models the relationship with γ_i , μ_i is an individual fixed effect, and ε_i^b is an error term. In this case, we also introduce treatment-fixed effects to account for unobserved individual heterogeneity.

In this regression, β^{good} is the coefficient of interest. It represents the effect of the message associated with the good on the degree of marginal positionality. In all cases, the comparison of the parameter of positionality is against the income positionality (the parameter elicited in the relative income experiment when the participants receive the baseline instructions). For example, the case of the β^{jewelry} reflects the extra positionality concerns associated with the jewelry's relative consumption.

Because our empirical strategy only allows us to recover a range for the implied γ for income and each good, our estimation strategy requires further assumptions about its distribution within each interval. Our baseline model estimate equation (6) is based on Random Effect and Fixed Effect model, which assumes that γ is uniformly distributed within each interval.⁶ As an alternative, we estimate these equations using maximum likelihood and an interval regressions model. The assumption, in this case, is that γ is distributed normally within each interval. As a complementary strategy, we also report the distribution of frequencies by good taking as reference results provided by the relative income experiment. We use the Kolmogorov-Smirnov test of equality of the two distributions to explore differences in the degree of positionality by type of good.

To test hypothesis H2, we explore the conditioned distribution of γ for each good. We created transition matrices to check whether the individuals' degree of positional concern is sensitive to the relative income and goods' consumption information.⁷

We use some standard persistence and transition rate measures to summarize this information. A persistence index was calculated by summing the unconditioned frequencies of the matrix's main diagonal. As an additional measure, we estimate the share of individuals with γ^{goods} higher than γ^{income} (the cumulative frequency of individuals located in the main diagonal and the cells above the main diagonal). Both summary indices measure the frequency with which the proposed hypothesis is fulfilled but do not allow statistical inference. For this purpose, we use the Kolmogorov-Smirnov test to compare conditional distributions. Given this objective and the number of observations, we conditioned the distribution of each γ^{goods} on two groups according to the magnitude of γ^{income} . The first group considers individuals with low-income positionality ($\gamma^{\text{income}} \leq 0.5$), while the second group considers those with high-income positionality ($\gamma^{\text{income}} \geq 0.5$). Kolmogorov-Smirnov test assesses the hypothesis that γ^{goods} for group 1

⁶For participants who choose society A over C_1 , we can only say that $-\infty < \gamma \le -0.125$. Analogously, for participants who choose C_5 over $A, -\infty > \gamma \ge 1$. For the first group, we use $\gamma = 0$, corresponding to the interval's upper bound. For the second group, we use the sum of the lower bound (1) and the length of the widest interval (0.25 = 1.25 - 1.00).

⁷As individuals were exposed to the experimental survey following the same sequence, concerns about potential order effects interpret transition matrices are less clear in this case. This point is addressed in the robustness check section.

(group 2) contains smaller values than for group 2 (group 1). Also, it assesses the null hypothesis of equal conditioned distribution.

Finally, to test hypothesis H3-a and to explore the individual's heterogeneity in the positional degree by sex of the respondents, we adapted the specification of the equation (6). We exclude the fixed effect and incorporate a wide set of control variables. Additionally, we test whether significant differences exist in the γ distribution of income/goods between males and females and those treated with a granddaughter hypothetical scenario versus a grandson hypothetical scenario. We also tested for differences in the distributions. Since the samples were independent, we used the Epps-Singleton test, which performs better than the Kolmogorov-Smirnov test for identifying differences in these distributions.

Unlike our previous hypotheses, our test for H3-b and H3-c compares the income positionality parameter between the control and treatment groups. Our control group is based on the responses of relative income parameters with baseline instruction. The treatment groups are based on the relative income experiment when participants know that inequality is mostly associated with inheritance or a differential lifelong effort. This specification allows us to estimate each treatment arm's effect using the control group as the comparison group. Since the only difference between the two is the additional information shown to the treatment group, our results can be interpreted as the effect of the additional message on the marginal degree of positionality parameter.

$$\gamma_i^{\text{income}} = \Gamma(\alpha + \beta^u \cdot IS_i^u + \delta X_i + \varepsilon_i) \tag{7}$$

In this case, the outcome variable $(\gamma_i^{\text{income}})$ represents the marginal income degree of positionality. IS_i^u are dummy variables indicating whether the choice of participant *i* was made, knowing that inequality is mostly associated with inheritance or a differential lifelong effort (*u*). Note that β^u can be interpreted as the effect of information about the source of inequality on the income concern parameters.

4 Data and Implementation

4.1 Data

The sample used in this experiment comes from a sub-sample of the Longitudinal Welfare Study in Uruguay (*Estudio Longitudinal del Bienestar en Uruguay*, ELBU by its Spanish acronym). ELBU is a survey with four waves, which started in 2004 and interviewed state primary school pupils in their first year. Its objective was to produce information on different socioeconomic dimensions in the context of social crises in Uruguay. The last wave was completed in 2015 - 2016 and from a total sample of 816 ELBU youngsters surveyed in Montevideo (the capital city of Uruguay). For our experiment in 2018, we successfully recontacted and visited 554 of them (Sample A). Rejections or inability to could recontact youngsters (using the address and telephone we had from the survey) explain the difference of 262 who did not join in the experiment to participate. In this sample, we apply the grandchildren's gender information treatment.

We asked the 554 participants to refer a friend willing to participate in the experimental survey. We

obtained contact information and visited 397 of those contacts (Sample B). The difference between the number of ELBU and the new contacts is explained mainly because the referred friend was not available to participate in the experimental survey, and to a smaller extent, because of errors in the contact information provided by the ELBU youngsters. We apply the fairness (effort or inheritance) information treatment in this sample. Additionally, in this sample, we do not apply the hypothetical scenario associated with the car.

This leads to a total sample of 951 youngsters that complete the experimental survey. The questionnaire was completed individually in the respondent's home with the presence of an interviewer. This would avoid contagion effect in the responses. To motivate interviewees to complete the experiment, the participants were offered the chance to win a USD 500 prize. This potential prize is independent of the respondent's choices, and the lottery includes all participants.

4.2 Final samples and randomization groups

In this section, we address two additional concerns regarding our data: the consistency of participants' responses in the experiment and the balance of our information treatment.

It is worth noting that the way in which we elicit individuals' marginal positionality parameter with this strategy implies that –if individuals are rational– once they stop choosing C_z and start choosing A, they should not go back to C_{z+1} ever again (weakly monotonic preferences assumption). The inconsistent responses could be explained by problems of comprehension, low attention, fatigue of the participants, or the restrictive assumption of weakly monotonic preferences and the functional form of the utility function.⁸

If the inconsistent responses are associated with certain participant characteristics, it could yield bias in the estimated parameters. To address this issue, we use alternative criteria of consistent responses and identify alternative groups based on the consistency of responses between experiments. First, the condition we use to define our Baseline Criterion (BC) is to drop respondents' information with inconsistent choice patterns. We also use a more flexible definition of consistent responses, which allows an expansion of the number of responses. We incorporate a simple assumption to recover some responses. Table A3 in the Appendix presents the criteria used to recover these cases (basically, we recover the participants who perform a single inconsistent responses by good/income for the two criteria defined and for each sample (A, B, and entire). The proportion of consistent responses is similar between goods and higher when income is considered. There are no differences in the proportion of consistent responses between Samples A and B, either when we use the Baseline or Extended Criterion (Panel I and II), or when we consider the number of experiments in which participants provide consistent responses (Group

⁸Bergolo et al. (2022) use an online experimental survey for university students to elicit inequality aversion parameters. They explore some sources of inconsistent responses and implement a wide range of attention and comprehension checks. They found that female participants were, on average, more likely to be inconsistent than male participants. They suggest that the participants' fatigue is not related to inconsistent responses. They carried out estimates restricting the sample to those who reported having paid attention and answered the comprehension check correctly, and the main results remained essentially the same. Their main results are robust to alternative ways of handling inconsistent responses. Their results remain when they use alternative assumptions about the utility function.

I and Group II).

Because respondents may be consistent in one experiment and not in the others, we identify two groups of responses. The first group includes consistent responses for each experiment. Namely, it includes respondents that make consistent choices for the sequences of at least one of the experiments – for income or each good (Group I). The second group is defined in the most restrictive way: we exclude participants that make inconsistent choices in at least one of the sequences of the experiments – consistent in every goods/income experiment – (Group II). This is a very restrictive group since it excludes information from respondents who were consistent in three of the experiment but inconsistent in a fourth. In this case, we exclude the car because the hypothetical scenario was only applied to Sample A. While Group I focuses on experiments with consistent responses, Group II focuses on the consistency of the participants.

It is pertinent to mention that the final sample of respondents is substantially reduced when we apply the most restrictive criteria. Considering that the experiment was conducted face-to-face, the percentage of inconsistent responses is relatively high in relation to previous similar studies. The most similar paper was Alpizar et al. (2005), which has a 13.5% of inconsistent responses. Some aspects could explain this difference. First, the number of consistent choices in the four choice sequences is higher than in the previous studies (in Alpizar et al. (2005) respondents make 12 choices between hypothetical societies, while in our experiments, they make 20 choices). The levels of consistency are much higher and comparable when we consider each experiment separately. This is particularly the case for income. Most of the previous comparable studies use samples of university students interviewed in the classroom, unlike our study, whose participants are young people with different educational levels interviewed face-to-face at their homes. The results are presented for the four alternatives mentioned to identify whether inconsistent responses generate any bias in our estimates.

				Sample	
			А	В	Entire
			N=554	N=397	N=951
	Group I: participants con	sistent	t in at least	one exper	iment
	Consistent with Income	Ν	363	260	623
	Consistent with Income	%	0.655	0.655	0.655
	Consistent with Iswalm	Ν	231	157	388
Danal I. Deseline	Consistent with Jewelry	%	0.417	0.394	0.408
Critoria (PC) of	Consistent with Health	Ν	274	204	478
cinteria (BC) or	Insurance	%	0.495	0.514	0.503
consistency	Consistent with Con	Ν	252		
	Consistent with Car	%	0.455		
	Group II: participants co	nsister	nt in every	experimen	t †
	Excluding cor	Ν	112	83	195
	Excluding car	%	0.202	0.209	0.205
	Including our	Ν	82		
	including car	%	0.148		
	Group I: participants con	sistent	t in at least	one exper	iment
	Consistent with Income	Ν	459	329	788
	Consistent with Income	%	0.829	0.829	0.829
Denal II. Enter de d		Ν	396	266	662
Panel II. Extended	Consistent with Jeweiry	%	0.715	0.670	0.696
Criterion (EC) of	Consistent with Health	Ν	428	314	742
consistency	Insurance	%	0.773	0.791	0.780
	Complete and south Com	Ν	417		
	Consistent with Car	%	0.753		
	Group II: participants co	nsister	nt in every	experimen	t †
	Evoluting our	Ν	283	195	478
	Excluding car	%	0.511	0.491	0.503
	Including our	Ν	239		
	menuting car	%	0.431		

Table 1: Different types of consistency for each sample

Sample A comprises young people who participate in ELBU, while sample B comprises contacts from friends provided by ELBU's young people. The difference in the number of cases is because the friends were not located or refused to participate in the experiment. The entire sample arises from the sum of samples A and B. † We exclude the car experiment responses in Group II because the relative consumption experiment for this good was only applied to Sample A.

In addition, we have information to evaluate whether the inconsistency of the responses is associated with participant characteristics. The survey includes a set of demographic (sex, age) and socioeconomic variables (household size, employment, education level, emancipation – young people who live in a different house from their parents –, and parental educational achievement). Table 2 presents the mean for these variables in Group I (jewelry, health insurance, and income) and Group II (always consistent) with the alternative consistency criteria (BC and EC). Results suggest that consistent and inconsistent groups are generally balanced in observable characteristics, and particularly when the sample size increases and the entire sample and the extended criterion are considered. The only significant differences appear when considering young people emancipated from their home of origin in Group II and male and educational level in Group II with Baseline Criterion. In Table A4 of Appendix, we show the mean of Sample A with Group I. In this case, we include the γ parameter of the car and test the difference between this mean and the mean of the other goods and income. In these cases, also we did not find significant differences.

			(Group I				Group	II
	Jewelry	Insurance	Income		p-value		Yes	No	p-value
	(1)	(2)	(3)	(1) vs. (2)	(1) vs. (3)	(2) vs. (3)	(4)	(5)	(4) vs. (5)
I. Baseline Criterion									
Age	20.53	20.36	20.38	0.146	0.242	0.853	20.48	20.41	0.666
Male	0.517	0.488	0.488	0.409	0.373	0.986	0.559	0.457	0.011
Emancipated	0.181	0.178	0.176	0.919	0.829	0.908	0.221	0.146	0.012
Household size	3.974	3.827	3.851	0.209	0.271	0.817	3.840	3.923	0.525
Work (1=Yes)	0.386	0.419	0.415	0.322	0.371	0.874	0.426	0.398	0.478
< 10 years of education	0.342	0.300	0.277	0.187	0.029	0.408	0.349	0.279	0.053
II. Extended Criterion									
Age	20.50	20.41	20.41	0.408	0.419	0.982	20.51	20.35	0.233
Male	0.478	0.475	0.484	0.910	0.818	0.722	0.491	0.465	0.434
Emancipated	0.182	0.160	0.172	0.266	0.616	0.519	0.189	0.133	0.019
Household size	3.912	3.906	3.880	0.866	0.653	0.773	3.866	3.953	0.435
Work (1=Yes)	0.395	0.402	0.417	0.781	0.398	0.562	0.404	0.403	0.979
< 10 years of education	0.311	0.291	0.290	0.411	0.364	0.939	0.319	0.268	0.105

Table 2: Mean of demographic and socioeconomic variables. Different samples and consistency criteria.

 Entire sample

Descriptive statistics based on the entire sample of participants (Sample A + Sample B: N=951). The Baseline Criterion includes responses with strict consistency, while the Extended Criterion uses a more flexible definition of consistency, adding responses with only one inconsistency to the sequence of responses. Group I considers the sample of responses that are consistent in each good, while Group II considers the sample of participants that provide consistent responses across all goods.

Finally, Tables A6 and A7 in the Appendix allow us to compare the balance in the characteristics between participants assigned to different information treatment groups. In general, results confirm that the randomized groups are also balanced in observable characteristics. Our treatment arms thus identify the causal effect of this information treatment on the magnitude of the parameter of marginal positionality (γ).

5 Main results

5.1 Baseline estimate for positional concern

This section reports the average marginal degree of positionality (γ) for income and the aforementioned three goods. Table 3 shows results based on the two alternative notions of consistency of individuals' responses: Baseline and Extended Criterion. Also, we present the results for the two groups of respondents: participants' responses were consistent in at least one experiment (Group I); responses of participants that made consistent choices in the four sequences of the experiments (Group II). Results show that the marginal degree of the positionality of income is between 0.55 and 0.65, depending on the criteria used to define the consistency. The average degree of positionality is also positive and of greater magnitude than income for health insurance (0.76 - 0.91), jewelry (0.82 - 0.90), and car (0.85 - 0.92). As is expected, the average magnitude is slightly higher when considering the extended criterion. A means test was conducted to check whether the differences were statistically significant. In all cases, the t-student test of equality of the average marginal degree of positionality between income and goods is rejected at a 1% significance level. The magnitude of the average marginal degree of positionality between income and goods is rejected goods is very close.

	I. E	I. Baseline Criterion			П. І	Extende	d Criteri	on	
	Grou	ıp I	Grou	Group II Grou		ıp I	Grou	Group II	
	γ	N	γ	N	γ	Ν	γ	Ν	
Health Insurance	0.82	478	0.76	195	0.91	742	0.91	478	
Jewelry	0.82	388	0.83	195	0.90	662	0.90	478	
Car [†]	0.85	252	0.88	82	0.90	417	0.92	239	
Income	0.63	623	0.55	195	0.65	788	0.62	478	
T-test with consister	nt case:	γ^{goods} -	$-\gamma^{\text{income}}$	= 0					
Good: Health Insura	ance	3.843					8.62	26	
Good: Jewelry			5.1	11			8.27	74	
Good: Car (Sample	A)		4.10	00			6.52	24	

Table 3: Average marginal degree of positionality. Entire sample

The Baseline Criterion includes responses with strict consistency, while the Extended Criterion uses a more flexible definition of consistency, adding responses with only one inconsistency to the sequence of responses. Group I considers the sample of responses that are consistent in each good, while Group II considers the sample of participants that provide consistent responses across all goods. [†] The hypothetical scenario of the cars was only presented to sample A. Therefore, consistency refers to that population.

A more demanding strategy to test the H1-b hypothesis is presented in Table 4, which reports the results of our parametric estimates for both criteria. Columns (1) and (2) report the Random Effect model, Columns (3) and (4) report the Fixed Effect model, while (6) and (7) report the estimates based on Interval Regressions. In columns (1), (2), (6), and (7), we include a set of control variables. These controls are a set of individual characteristics of the participants (demographic and socioeconomic background). In each case, we report the results for all cases (Group I) and the results based on consistent responses across all goods/income experiments (Group II).

Our preferred specification is reported in Columns 3 and 4 because we control for the influence of unobservable variables at the individual level. As a result, these specifications capture the variation in γ^{goods} and γ^{income} at the individual level and report the average difference. They support a more causal interpretation of our results.⁹ In the specifications, we omit the γ of the income. Among other things, the Fixed Effect allows us to approximate the magnitude of this γ with greater precision using the coefficient of the constant. In this sense, estimates based on this last model confirm a positive and statistically significant marginal degree of the positionality of income, and its magnitude is around 0.60 (constant coefficient). The effect of selected goods is statistically significant at the 1% significance level, and their signs are positive. According to this parameter, the positionality of jewelry is between 22% and 30% higher than the positionality of income. Similar results are obtained for cars (23% and 26%) and health insurance (20% and 27%). The differences observed when comparing the marginal degree of positionality parameter of jewelry and income are similar in magnitude to the differences observed when comparing health insurance and income, and cars and income, respectively.

The estimation of the income positionality parameter using the constant coefficient is less precise in the Random Effect model (Columns 1 and 2) and Interval Regression (Columns 5 and 6). Although the coefficient is positive, it is not always significant. Its magnitude varies considerably since it captures unobservable or omitted information. However, the coefficients for the goods treatment are consistently

 $^{^{9}}$ A limitation to this interpretation could be the order effect. However, the type of questionnaire, the sequence of choices, and the tests performed suggest this would not be a very relevant problem in this context. A more formal empirical analysis of the potential effects of response order is presented in Section 6.

positive across different specifications and criteria, indicating that unobservable characteristics of the goods explain some of the differences in individuals' positionality. In all models, the treatment effects for all goods are statistically significant at the 1% significance level and have the expected sign. The coefficients of the interval regression model (Columns 5 and 6) are much larger than those of the other models because this estimation is more sensitive to extreme values in the distribution. When we exclude observations at both extremes of the distribution in the interval regression, the coefficients have a magnitude comparable to those of the other models. We will present this result in Section 6.

	Random Effect		Fixed Effect		Int. Reg.	
I. Baseline Criterion						
Omitted variable: Income						
Health Insurance	0.205***	0.221***	0.212***	0.222***	0.527***	0.769***
	(0.032)	(0.056)	(0.037)	(0.056)	(0.084)	(0.215)
Jewelry	0.205***	0.298***	0.222***	0.298***	0.542***	1.132***
	(0.035)	(0.055)	(0.039)	(0.055)	(0.091)	(0.224)
Car	0.222***		0.234***		0.549***	
	(0.038)		(0.041)		(0.108)	
Constant	1.166***	0.214	0.614***	0.539***	2.328	-1.562
	(0.404)	(2.207)	(0.020)	(0.034)	(1.519)	(8.315)
Observations	1,638	561	1,638	561	1,638	561
R2	0.062	0.089	0.062	0.089		
II. Extended Criterion						
Omitted variable: Income						
Health Insurance	0.266***	0.282***	0.267***	0.282***	1.003***	1.138***
	(0.026)	(0.034)	(0.028)	(0.034)	(0.105)	(0.152)
Jewelry	0.247***	0.283***	0.251***	0.283***	0.947***	1.166***
-	(0.029)	(0.035)	(0.030)	(0.035)	(0.109)	(0.153)
Car	0.252***		0.250***		0.912***	
	(0.032)		(0.033)		(0.129)	
Constant	1.113***	0.730*	0.645***	0.621***	3.444*	3.078
	(0.318)	(0.424)	(0.017)	(0.021)	(2.017)	(2.928)
Observations	2,436	1,350	2,436	1,350	2,436	1,350
R2	0.077	0.099	0.078	0.099		
Group	Ι	II	Ι	II	I	II
Controls	Yes	Yes	No	No	Yes	Yes

Table 4: γ estimation. Different Specifications and consistency criteria

The covariates included as controls are: age; age squared; sex (1=female); household size; if the young person left the home of origin (1=Yes); years of education (1=<10 years of education); and labor condition based on two dummy variables (work between 30 and 40 hours a week, and work more than 40 hours a week). Additionally, we include a dummy variable that identifies whether the young person belongs to sample A or B. *Significant at 10%, **significant at 5%, ***significant at 1%. Standard errors are in parentheses.

Indeed, results for our sample of Uruguayan youngsters are comparable with evidence for students from Costa Rica reached by Alpizar et al. (2005). The results in the Appendix in Figure A2 show that the average γ for income and each good are similar. However, in our case, the magnitudes are a bit higher.¹⁰

One novel finding of our results is that although health insurance is not socially visible, its marginal degree of positionality is relatively high and comparable in magnitude with the degree found for cars and jewelry. In addition, its magnitude is two times higher than the positionality of insurance reported by Alpizar et al. (2005). This result suggests that positional concern is not directly associated with the social

¹⁰Note that the list of goods of both studies is not entirely comparable, and the intervals represented by the societies in both questionnaires are not the same. In particular, there is an overlap for some intervals, but our experimental survey includes a greater number of intervals. Furthermore, in the relative consumption experiment by Alpizar et al. (2005), participants are asked to choose among three pairs of societies, while in the case of the relative income experiment, they choose between seven pairs of societies. In our case, participants chose between five pairs of societies in all cases.

visibility of expenditures. We found three possible interpretations to explain the relatively high degree of the positionality of Health Insurance. First, it could be associated with status-seeking behavior, given that access to prime health insurance could be associated with higher wealth and successful life. Second, health may be perceived as a merit good and motivate higher relative consumption. Third, it could be related to the information effect. In this case, the participants could perceive the relative expenditure on health services as a signal of their grandchild's health and the quality of access to health care services. Although it is beyond the scope of this paper to provide evidence about these hypotheses, we explore whether the magnitude of $\gamma^{\text{insurance}}$ is related to their employment status, the relevance of health in their lives, and the respondents' assessment of public health insurance. Table A8 in the Appendix presents the average marginal degree of positionality for alternative samples, criteria, and groups. It shows that respondents who negatively evaluate public health insurance, who are not employed (less likely to have access to private insurance), and those who assign low priority to health have a higher positionality in health insurance. These results are consistent with the three arguments put forward to explain the high value of $\gamma^{\text{insurance}}$. However, significant differences are only confirmed when we consider differences in the respondents' employment status. The lack of significance in the rest of the tests may be due to the low power derived from the low number of observations. Therefore, based on this evidence, we cannot rule out that the difference is motivated by the search for the status of non-employees.

Our results are consistent with hypothesis H1-a, although they reject the difference. The effects are consistently positive across the different specifications, consistency criteria, groups, and samples, showing a greater average degree of positionality for jewelry, health insurance, and cars. These results are consistent with hypothesis H1-b.

5.2 It is the positionality and individual's attribute?

The previous section focused on the average results, and in this section, we will delve deeper into the distributions to explore the existence of heterogeneities between individuals. Figure 1 shows the γ^{income} distribution for the set of experiments. Panels (a), (b), and (c) compare the distribution of γ^{income} with the distribution of γ^{income} , γ^{jewelry} and γ^{car} respectively. Figure A3 in the Appendix presents the analogous results for the Extended Criterion. On the *x*-axis we report the implied value of $\gamma^{\text{income}}/\gamma^{\text{good}}$ associated with different alternative choices of A and C_z and on the *y*-axis we report the frequency of $\gamma^{\text{income}}/\gamma^{\text{good}}$.

The distribution of γ^{income} reveals several findings. First, for most subjects, relative income matters: the marginal degree of positionality for income for the median participant belongs to the interval [0.5, 0.75), and the estimate of the mean is around 0.6. This means that, on average, an increase of \$1 in subjects' income generates more utility via the increase in their relative income than it does an increase in the absolute income. Second, the γ^{income} distribution is bimodal and forms a U-shaped curve. This functional form accords with previous findings in the literature (see, for instance, Alpizar et al. (2005)). Third, it is also worth noting that between 26% and 32% of the participants consider income not a positional good. Finally, between 38% and 42% of participants present a $\gamma^{\text{income}} \ge 0.5$. They could be identified defined as extremely positional subjects.

Figure 1 confirms the same U-shape pattern for the marginal degree of positionality for health insurance,

jewelry, and cars. The results reported in panels (a), (b), and (c) are powerful: changing the hypothetical scenario from the income to the consumption of the selected goods noticeably shifts the distribution of the marginal degree of positional parameter toward the right concerning the distribution of γ^{income} . The three goods accumulate more than 50% of the frequencies in the range $\gamma^{\text{goods}} \ge 1$, suggesting that most participants are highly positional concerning these goods. In the opposite extreme, between 20% and 24% of the participants (depending on the good) are located in the range $\gamma^{\text{goods}} \le 0$, namely, they are positional-neutral. These levels are higher than the frequency obtained for income (between 26% - 38%). The K-S test of equality of the two distributions is rejected at a 1% significance level in all cases. These results remain when we use the Extended Criterion (see Figure A3 in the Appendix) and confirm the previous conclusion regarding the hypotheses H2.

Figure 1: Marginal degree of positionality distribution - alternatives goods. Baseline Criterion



This figure displays the distribution of γ estimated using the Baseline Criterion in each case. In the *x*-axis, we report the implied value of γ associated with different choices of *A* and *C_z*. On the *y*-axis, we report the frequency of γ associated with each good and income. The dot-dashed line represents our estimate for the mean using OLS regression of γ over a constant. p-values Kolmogorov-Smirnov tests for equal distribution are presented in each graph.

An interesting point is whether the degree of the positionality of each individual persists for income and different goods. Our baseline H2-a establishes that those individuals with higher income positional concerns have a higher marginal degree of positionality for the analyzed goods. An alternative hypothesis is that respondents have high positionality for income but not for cars, jewelry, or health insurance. To address this hypothesis, we analyze transition matrices of the individual marginal degrees of positionality in the context of our within-individual experiment. The transition matrix describes the individual's probability of a change from one state, the range of γ^{income} to another state, γ^{good} . We also explore the individuals' γ^{good} transition between the selected goods.

In panels (a), (b), and (c) of Figure 2, the y-axis represents the range of γ^{income} and the x-axis represent the range of $\gamma^{\text{insurance}}$, γ^{jewelry} and γ^{car} . The diagonal of the matrix represents the persistence of subjects in each of the six ranges of γ for the income and set of goods considered. Namely, respondents with the same range of γ for the two endowments considered in the matrix. When we use the Baseline criterion of consistency, the persistence indexes are 31%, 31%, and 37% for jewelry, health insurance, and cars, respectively (see the first row and three columns in Table 5). The result is similar when the extended criterion is used (Cols. 3, 4, and 5 in Table 5). In other words, many participants have the same positionality level for income and goods analyzed.

Overall, the dynamics of the positional parameter are consistent with our previous analysis. The bimodal

distribution is confirmed, with a high concentration at the extremes of the matrix diagonal. A significant proportion of respondents have a high degree of positionality for all goods and income. And a smaller percentage is positional-neutral for the goods analyzed. However, the consistency in positionality between income and goods is not perfect. The degree of the positionality of the goods (x-axis) tends to be higher when the positionality of income is higher. Additionally, the shares of respondents that present an equal or higher γ^{good} than γ^{income} are 73%, 77%, and 80% for jewelry, health insurance, and cars, respectively (see the second row and the first three columns in Table 5). This confirms what the matrices suggested and implies that the most frequent alternative in these samples is consistent with hypothesis H2.

K-S tests were conducted to check whether the differences in conditional distribution are statistically significant. To construct the conditional distributions, we divided the responses into two groups: low-income positionality individuals (when $\gamma^{\text{income}} \leq 0.5$); high-income positionality individuals (when $\gamma^{\text{income}} > 0.5$). Then we explore the γ^{goods} distribution for each group. The results are summarized in the last three rows of Table 5). We find favorable evidence for hypothesis H2-a in the cases of health insurance and cars. In the case of cars, the tests reject that the distributions of both these groups are equal and that the magnitude of γ^{car} is smaller for the group of low-income positionality individuals than for the groups of high-income positionality individuals. These results are confirmed for Baseline Criterion (Col. 3) and the Extended Criterion (Col. 6). The same conclusion is reached for hypothesis H2-a for health insurance. However, the weaker significance is only confirmed for the Extended Criterion. This could be associated with the sample size. Although these results are consistent with our hypothesis, it is worth emphasizing the limits of this exercise. As individuals were exposed to alternative consumption and relative income experiments with the same sequence, concerns about potential order effects led us to consider transition matrices cautiously. Furthermore, γ^{good} tends to become more concentrated in the high-value ranges, consistent with the previous hypothesis.

As a complementary analysis, panels (a), (b), and (c) of Figure A4 in the Appendix compare the analogous transition between goods while Table A9 displays the synthetic index and the conditional distribution tests. On the one hand, this analysis makes it possible to evaluate the persistence of individuals' positionality among alternative goods. On the other hand, it helps to explore whether one is more positional than another. Again, we confirm the heterogeneity in the ranges of γ at the individual levels between goods. However, a greater accumulation of frequencies is observed on the main diagonal, primarily in the cases of jewelry vs. health insurance and car vs. health insurance. The results based on K-S tests suggest a strong association in the degree of positionality between goods. That is, when an individual has a high positionality in health insurance (or car), he or she tends to have a higher γ^{car} and $\gamma^{jewerly}$ (or $\gamma^{jewerly}$).



Figure 2: Matrix transition of subjects a marginal degree of positionality. Baseline Criterion

This figure displays subjects' movements (and their γ^b) when they choose alternative treatments. We created these transition matrices to represent the transition probability between an income concern parameter and marginal positionality parameter for health insurance, cars, and jewelry (panels a, b, and c, respectively). Each row in the matrix represents the γ^{income} under the relative income experiment, while each column represents the γ^{good} under the relative, conditional to the individual's γ^{income} . The diagonal of the matrix represents the persistence of subjects in each of the six ranges of γ for the income and goods considered.

Table 5: Synthetic index and conditional distribution tests. Conditional variable: income. Entire sample

	Bas	eline Criterio	n	Exte	ended Criterio	n
Goods:	Jewerly	Insurance	Car	Jewerly	Insurance	Car
Transition matrix						
% persistence ($\gamma^{\text{goods}} = \gamma^{\text{income}}$)	0.31	0.31	0.37	0.34	0.36	0.37
$\% \gamma^{\text{goods}} \ge \gamma^{\text{income}}$	0.73	0.77	0.80	0.78	0.83	0.82
Conditional distribution test: p - value						
$H_0: \gamma_i^{\text{goods}} < \gamma_i^{\text{goods}} \text{ if } \gamma_i^{\text{income}} \le 0.5 \text{ and } \gamma_i^{\text{income}} > 0.5$	0.23	0.28	0.00	0.29	0.03	0.02
$H_0: \gamma_i^{\text{goods}} > \gamma_i^{\text{goods}}$ if $\gamma_i^{\text{income}} \le 0.5$ and $\gamma_i^{\text{income}} > 0.5$	1.00	1.00	1.00	1.00	1.00	1.00
$H_0: F_i(\gamma^{\text{goods}} \mid \gamma^{\text{income}} \le 0.5) = F_j(\gamma^{\text{goods}} \mid \gamma^{\text{income}} > 0.5)$	0.41	0.51	0.01	0.53	0.05	0.04
Ν	297	353	194	567	639	358

This Table displays the synthetic measures from the transition matrix and the test of differences on the conditional distributions $F_i(\gamma^{goods} \mid \gamma_u^{income} \le 0.5)$ and $F_j(\gamma^{goods} \mid \gamma_d^{income}) > 0.5$. We report the frequency of persistence between γ^{goods} and γ^{income} (the sum of the unconditioned frequencies of the matrix's main diagonal). p-values Kolmogorov-Smirnov tests for the three alternative hypotheses. The p-value of the equal distribution test is estimated using the exact option.

5.3 Some drivers of positional concern

In this subsection, first, we explore the presence of differences in the positionality parameter by gender. Second, we assess the role of luck and inheritance in determining the positional level of participants. These subsections allow us to test hypotheses H3-a and H3-b/H3-c, respectively.

5.3.1 Does the gender of respondents and their grandchild matter?

Gender is a source of heterogeneity in risk attitudes, attitudes towards competition, and social preferences (Bertrand, 2011). Walker and Frank (1999) states that men should be more positional concerned based on the evolutionary argument. We replicate the analysis in Figure 1, and compare the γ^b distribution and their means for men and females. We conducted mean and Epps-Singleton tests to compare the results and found no statistical differences between gender. We also confirm the same U-shape pattern for the marginal degree of positionality (see Figure A5 in the Appendix).

The previous analysis focused on the differences between the γ^b distribution by gender. Now, we shift our focus to the average marginal degree of positionality. Referring back to the estimate in Table 4, we present the complete estimate for the Random Effects and Interval Regression model in Table A5 of the Appendix. One of the covariates used in the model is gender, which is statistically significant and positive when considering the Extended Criterion.¹¹ Our analyses suggest no significant difference by gender, even if there is weak evidence that females have higher positionality than men. These results contradict the evolutionary arguments on which our hypothesis (H3-a) is based on that positionality and with the results obtained by Johansson-Stenman et al. (2002). However, Alpizar et al. (2005) also confirmed a greater relative concern for females than for men and suggested that the differences would be driven by fairness concerns because females have more socially oriented behavior.

As an additional exercise, we exploit grandchildren's gender information treatment. Respondents may have gender preferences for their grandchildren and different levels of altruism for the next generation. In our relative income and relative consumption experiment, we introduce two original information treatments that alternatively considers the well-being of their imagined granddaughter or imagined grandson. Because the sex of the grandchild was randomly distributed, we estimate the treatment effect on the mean of the γ distribution. We replicate the analysis presented in Figure A5 but by grandchild gender. Figure A6 in the Appendix shows these results and suggests no significant differences in the parameter between responses based on granddaughters and grandsons. We also compare results based on these treatment groups with the control group (text said grandson/granddaughter), and we do not find statistically significant differences.

5.3.2 Positional concern and the role of the origin of inequality

In this section, we focus on fairness treatment to assess the contribution of self-centered income inequality to our measure of positional concern. This treatment was only applied to Sample B and the relative income experiment. Table 6 reports the results of our parametric estimates for the Baseline and Extended Criterion. Columns (1) to (2) report the result of the Random Effect model, columns (3) and (4) the Fixed Effect model, while (6) and (7) report the estimates based on interval regressions. Columns (1), (2), (6), and (7) include the same set of control variables as Table 4. As Table 4, we report the results for all cases and with consistency criterion, but in this case, only between income's hypothetical scenario and scenario with fairness informational treatment (Income-treatments consistency).

The differences observed when comparing the marginal positionality parameter for income of the *in-heritance* and control groups are negative in all cases, but their significance is weak. Only in the cases of Random Effect and Interval Regression models carried on the Balanced Criterion are the treatment effects of the *inheritance* message statistically significant at the 10% significance level. However, the non-significance of treatment seems to be associated with the sample size and a test power problem. The test power problem is alleviated when we use the Extended Criterion. In this case, the treatment effects of the *inheritance-message* are statistically significant at the 5% significance level for all regressions. The sign of the effect is consistently negative across the different specifications used, showing a lower degree of income positionality when participants receive a message framing that *inheritance* is the source of

¹¹The difference by gender is only confirmed for the average positionality. In order to assess whether the differences by gender are confirmed for any of the goods, we run additional estimates, including interaction terms between variables identifying the goods and females. In any case, we find significant differences. Therefore, we cannot attribute the higher positionality of females to the consumption of any specific goods. These results are not reported but are available from the authors.

inequality.

The effect of the *effort-message* is not statistically significant, although the sign of the effect is consistently negative across the different specifications and consistency criteria. When we consider these two information treatments together, their effect is negative and generally significant (see Table A10 in the Appendix).

These results, on the one hand, provide weak evidence regarding our hypothesis H3-b: $\gamma^{\text{income-effort}} \leq \gamma^{\text{income}}$. On the other hand, they support that $\gamma^{\text{income-inheritance}} < \gamma^{\text{income}}$. This result suggests that grand-children's inheritance is perceived as merit -the effort of the previous generation (parents/mothers and grandparents/grandmothers) and not as an unfair circumstance. Considering the hypothetical grandparents' response, the inheritance could be interpreted as the respondents' effort. This interpretation is in agreement with the idea that individuals are less likely to accept inequality when it comes from differential effort/inheritance, while they are more reluctant when it comes from circumstances that are beyond individual control as luck.

These results support the idea that the income positionality parameter is influenced by self-centered inequality aversion. One way to discount this effect is to calculate the difference between the positionality parameters of goods and income. As suggested by hypothesis H1-b, if a good is positional, the result of this difference should be positive. Note that this difference's magnitude establishes a lower positionality threshold motivated exclusively by consumption.

	Randor	n Effect	Fixed	Effect	Int. Reg.	
I. Baseline Criterion						
Omitted variable: Only income -	control grou	<i>p</i> -				
Effort	-0.051	-0.042	-0.048	-0.048	-0.089	-0.078
	(0.040)	(0.044)	(0.045)	(0.045)	(0.074)	(0.083)
Inheritance	-0.077*	-0.064	-0.058	-0.058	-0.133*	-0.124
	(0.039)	(0.043)	(0.044)	(0.045)	(0.070)	(0.079)
Constant	-0.052	0.023	0.622***	0.593***	-0.704	-0.443
	(0.485)	(0.595)	(0.010)	(0.016)	(1.014)	(1.397)
Observations	841	404	841	404	841	404
R2	0.014	0.014	0.014	0.014		
II. Extended Criterion						
Omitted variable: Only income -	control grou	<i>p</i> -				
Effort	-0.045	-0.046	-0.050	-0.050	-0.085	-0.085
	(0.040)	(0.042)	(0.044)	(0.044)	(0.087)	(0.088)
Inheritance	-0.098**	-0.093**	-0.089**	-0.089**	-0.195**	-0.183**
	(0.039)	(0.042)	(0.044)	(0.044)	(0.083)	(0.085)
Constant	-0.030	-0.099	0.652***	0.651***	-0.871	-0.761
	(0.467)	(0.549)	(0.009)	(0.016)	(1.185)	(1.379)
Observations	1,010	502	1,010	502	1,010	502
R2	0.022	0.022	0.022	0.022		
Controls	Yes	Yes	No	No	Yes	Yes
Income-treatments consistency	No	Yes	No	Yes	No	Yes

Table 6: Effort and Inheritance. Different Specifications and consistency criteria. Sample B

The covariates included as controls are: age; age squared; sex (1=female); household size; if the young person left the home of origin (1=Yes); years of education (1=<10 years of education); and labor condition based on two dummy variables (work between 30 and 40 hours a week, and work more than 40 hours a week). Additionally, we include a dummy variable that identifies whether the young person belongs to sample A or B. *Significant at 10%, **significant at 5%, ***significant at 1%. Standard errors are in parentheses.

6 Robustness checks and additional analysis

6.1 With what variables is the marginal positionality parameter associated?

Aiming to assess the validity of γ as a measure of the marginal degree of positionality, we analyze whether our estimates of this parameter correlate with variables related to the preferences for status. On the one hand, we identify whether respondents compare themselves with others by looking up or down based on the questions included in Leites et al. (2022). This question is available for the entire sample. We also use the questions proposed in Heffetz (2011) to identify the visibility of the goods. From the list of 31 goods, we selected the 10 with the highest visibility for the Uruguayan case (Leites et al., 2019) and calculated the percentage that each young person identifies with these goods. This question is only available for Sample A. Finally, we use a question available only for Sample B and refer to the goods that young people identify as being used by successful people and those that allow them to improve their social position. Three of the fifteen goods refer to jewelry (rings, necklaces, watches). We identify those that point to at least one of these goods. In Table A11 of the Appendix, we show descriptive statistics of preferences for status variables.

A multivariate linear regression model was used to analyze the relationship between the parameter of marginal positionality (γ) and the commented variables (see Table 7). The dependent variable is γ^{good} for each individual. We include a set of respondent characteristics and socioeconomic backgrounds as control variables: sex, age, hours worked, household size, educational achievement, and parental educational achievement. Given that we have information in a single sample (A or B) for most of the variables that approximate preferences for status, and the consequent low number of observations, we do not restrict the estimates to having simultaneously consistent responses in all goods and use the Extended Criterion to consider that the answers in each good are consistent.

The first panel includes the variables that reflect the direction of interpersonal comparisons (Veblen, 1994; Hirschman and Rothschild, 1973), distinguishing between upward-looking and downward-looking behavior. The specifications show that upward-looking individuals have a higher-than-average positionality while those downward-looking individuals have a lower-than-average positionality in the cases of jewelry and income.

The following two panels focus on a different dimension to explain positional concern: goods visibility and success signaling. The second panel presents the association between γ and the visibility index. A significant negative correlation exists between the visibility index and the income position parameter. This could be explained by the fact that individuals with high-income positionality (low visibility) tend to assign less weight to the visibility of goods consumed by others. On the other hand, in the third panel, we observe that those who think that jewelry identifies successful people or is a sign of improvements in social position have a lower parameter of positionality in health insurance. This result suggests that the magnitude of the positionality parameter is sensitive to the fairness view of the respondents. A plausible interpretation is that our positional concern measure includes some inequality aversion.

	(1)	(2)	(3)	(4)
	Jewelry	Car	Insurance	Income
I. Comparison with other p	eople. Enti	re Sample		
Looking up	0.118***	* 0.038	0.046	0.112***
	(0.045)	(0.054)	(0.041)	(0.038)
Looking down	-0.083*	-0.010	-0.053	-0.137**
	(0.045)	(0.058)	(0.043)	(0.038)
Obs.	623	382	699	732
R2	0.038	0.017	0.033	0.027
II. Visibility of good. Only	Sample A			
Visibility Index (ln)	0.318	-0.025	-0.021	-0.355**
	(0.212)	(0.196)	(0.187)	(0.175)
Obs.	365	382	392	414
R2	0.039	0.010	0.009	0.021
III. Consumption of jewels	ry (watch, c	hain, ring).	Only Sample	В
Successful people	-0.721* (0.436)		-0.820*** (0.296)	-0.426 (0.446)
Improves social position	-0.184 (0.301)		-0.627*** (0.241)	-0.098 (0.250)
Obs.	258		307	318
R2	0.029		0.096	0.033

Table 7: γ estimation and preference for status. Different Specifications. Extended Criterion. Estimates not consistent in all goods/income

Table A12 in the Appendix shows the full estimation of the entire sample. The first comment is that, with three exceptions, the demographic and socioeconomic background variables do not show a significant incidence. Young people who work between 30 and 40 hours have a higher positionality parameter for health insurance. On the other hand, women have a higher (albeit weaker) positionality parameter for jewelry. Associated with this parameter, it is also found that youth who live in households whose size is larger (generally the poorest) have a lower value.

6.2 Robustness checks

Desired number of children One concern is that respondents might provide responses motivated by 'moral satisfaction' (Kahneman and Knetsch, 1992), the desire to make a good impression on the experimenter (Gaertner and Schokkaert, 2012), signaling motives (Beshears et al., 2008), or 'self-image concerns' (Akerlof and Kranton, 2000). To mitigate these problems, our survey experiment follows the model of a previous paper and frames the experiment as a choice in which society participants prefer their hypothetical grandchild to live in 80 years, between hypothetical societies characterized by levels of average income and grandchild's income. This seeks to create distance between the choice and the current context. However, differences in the desired number of children across respondents may affect our measure of the marginal degree of positionality. A person who does not want children might respond differently than someone who strongly wants a large family. We have information about the desired number of children in our survey. That question is available in the fourth wave of ELBU. Roughly 67% of

The covariates included as controls are: age; age squared; sex (1=female); household size; if the young person left the home of origin (1=Yes); years of education (1=<10 years of education); and labor condition based on two dummy variables (work between 30 and 40 hours a week, and work more than 40 hours a week). Additionally, we include a dummy variable that identifies whether the young person belongs to sample A or B. *Significant at 10%, **significant at 5%, ***significant at 1%. Standard errors are in parentheses.

the participants in the experimental survey prefer not to have children (50% when we consider the entire sample of respondents).

Table A13 in the Appendix presents the rate of consistency for the alternative goods and criterion according to the desired number of children. We distinguish between those who declare they want children (cols. 2 and 4) and those who do not (cols. 1 and 3). There are no relevant differences in the consistency ratios of the responses. Table A14 in the Appendix presents the average marginal degree of positionality estimates by the desired number of children for the alternative groups and criteria. The results are consistent with the baseline results.

Risk preference We also reject the hypothesis that individuals' risk preferences drive the average marginal degree of positionality for income, jewelry, and cars. We identified individuals as risk-averse, risk-neutral, or risk-loving and found no significant differences among them in tests of means for the positionality parameter. However, in the case of health insurance, we observed differences: those who exhibit risk aversion have a significantly lower positionality parameter compared to those who are risk-neutral (see Table A15 of Appendix). This result is expected because risk-averse individuals prefer a society with less relative inequality in access to goods and are willing to pay more to reduce uncertainty in accessing good medical coverage.

A measure based on a random sample of one-shot choice The results presented are based on rigorous criteria regarding the consistency of the entire sequence of responses by the participants (Baseline Criterion) or because it requires consistency in all experiments (Group I).¹² As discussed in the previous section, all these criteria yield robust results but utilize a sub-sample in order to exclude inconsistent sequences of choices. To assess the robustness of our results and potential biases associated with inconsistent sequences, we use an alternative strategy based on each choice between pairs of hypothetical situations. This strategy assumes that errors are random and caused by participant distractions. Since unintentional errors cannot be identified, each choice is assumed to reflect the true preference over the available pair of alternatives. We treat each choice as independent within each sequence of choices, allowing us to consider choices within an inconsistent sequence. We assume that correct responses outweigh distracting errors, and therefore, if we draw individual choices from the set of all available choices instead of the entire sequence of choices, we can simulate an empirical frequency of preferences at the aggregate level. The strategy requires an additional assumption about how responses to a choice are distributed within a sequence of five choices. We consider two alternatives: In Simulation 1, the distribution is based on the empirical frequency of preferences arising from the choice draw. In Simulation 2, the alternatives are distributed equally at 50%. The procedure details are described in Appendix B.

Note that this strategy does not allow for eliciting the parameter at the individual level and only provides an aggregate approximation of the distribution. The main results are described in the Appendix in Figures A7. These figures simultaneously present the results from Simulation 1 and 2 and include the results based on the Baseline Criterion for the set of experiments as a benchmark. Panels (a), (b), and (c) compare the distribution of γ^{income} with the distribution of $\gamma^{\text{insurance}}$, γ^{jewelry} and γ^{car} respectively. The

¹²Additionally, the Extended Criterion requires correction for mistakes made by the participants. We incorporate some assumptions about the mistakes to recover the sequence.

distributions of $\gamma^{\text{insurance}}$, γ^{jewelry} and γ^{car} are bimodal, and the frequencies are concentrated at the lowest $(-\infty, 0)$ and highest $(1, +\infty)$ range of γ . However, γ^{income} presents more uniform distribution. For all goods and income, most people are positional. Finally, the results reported in the three panels are very strong: changing the relative situation of the individual from the income to the goods noticeably shifts the distribution of the parameter of positionality towards the right. Furthermore, the mean based on these simulations confirms that these goods are more positional than income. The average γ based on simulation 1 is 63 for Health insurance, 62 for Jewelry, 62 for Cars, and 49 for income. Although we could not make inferences from this strategy, the results are absolutely consistent with our main results regarding HI-a.

Lexicographic strategy A methodological concern is that respondents might provide a lexicographic strategy as a cognitively easy attempt to complete the survey. This could explain the large frequency of "extreme" values of γ . We follow (Alpizar et al., 2005) and analyze the impact of removing lexicographic responses on the mean marginal degree of positionality. The results remain qualitatively the same when we exclude the two extreme ranges of the distribution (Table A16 in the Appendix). We redid our main estimates by excluding both outliers (see Table A17 of Appendix). With the Baseline Criterion (Panel I), the estimates became more imprecise due to the low number of cases. However, when we considered the Extended Criterion (Panel II), the main results were maintained with only two slight changes. Firstly, the positionality of goods increased concerning income when we considered estimates with fixed and random effects. The most significant increase was observed in the case of health insurance. Secondly, the coefficients associated with the interval regression decreased by 30-40% compared to the base estimate. Both of these results indicate that the gaps between both estimates have significantly reduced.

Order effect The order in which the hypothetical scenarios are presented could affect the participants' responses due to fatigue or because they are learning how the experiment works. Unfortunately, it was impossible to randomize the order of the hypothetical scenarios for different goods and income during the fieldwork. In the entire sample, questions were first asked about the hypothetical scenario associated with jewelry, followed by the hypothetical scenario of health insurance. Lastly, the hypothetical income scenario was included. However, in Sample A, the hypothetical car scenario was included in the third position, which was not included in Sample B. Therefore, in Sample A, the income scenario was in the fourth position, while in Sample B, it was in the third. Table A18 in the Appendix tests the mean difference of the income positionality parameter in both samples and finds no significant differences, except in Group I of the Extended Criterion where there is a significant difference at 10%. Although this exercise is inconclusive, it indicates no effect of the order in which the scenarios were presented on the average marginal degree of income positionality.

Quantile regressions We also estimate the γ for each good using quantile regressions at the median.¹³ Our estimates based on quantile regressions are not affected by the specific values of γ at the distribution extremes, and the coefficients are very similar to interval regression. Table A19 of the Appendix presents the Baseline and Extended Criterion results. Our main results are robust across samples, and conclusions

¹³With this specification, we estimate the treatment effect on the median of the γ distribution instead of the effect on the mean as interval and Random/Fixed Effect estimation.

remain the same.

7 Final comments

This paper makes four main contributions to the literature on the field. First, it provides evidence for a developing country on the degree of the positionality of a list of goods with very different characteristics: health insurance (low visibility and high use value), jewelry (high visibility and low use value), and cars (high visibility and high use value). In addition, the degree of the positionality of income (low visibility and determinant of consumption capacity) is used as a benchmark. All goods present higher and more significant positionality than income. The latter presents a parameter of positionality significantly greater than zero. This result, on the one hand, confirms the findings of previous studies that associate positionality with the consumption of visible goods that signal status, economic success, and wealth.

On the other hand, a novel finding has emerged regarding the high degree of positionality observed for health insurance, suggesting that positional goods do not necessarily need to be visible. This is significant in terms of its implications for welfare and public policy. The level of positionality found for this group of goods suggests that consumption and income inequalities indirectly affect people's welfare. In the particular case of access to quality health services, it directly affects people's well-being due to its impact on health. But it would also have an indirect effect through positional concern and unequal access to this service.

Secondly, another novel result that requires further analysis is the low correlation between the positionality parameters and the variables on interpersonal comparisons and visibility of goods. In particular, we did not expect a non-significant correlation between the degree of income positionality and the relevance of income comparisons reported by individuals. These results are unexpected and raise some working hypotheses. On the one hand, the low correlation could indicate that the variables used incorporate measurement errors or measure different phenomena. This has methodological implications for understanding the origin of the difference between the measurements based on the experimental questionnaire (which assumes a specific functional form of relative concern) and those arising from direct questions. On the other hand, it could indicate that positionality is a phenomenon that encompasses multiple dimensions, and very different motivations can explain its origin depending on the goods considered, the groups taken as reference, the functional form of the relative concern, and the goods considered.

Third, the study provides evidence of the positionality parameter heterogeneity among individuals. The relatively low correlation in the positionality parameter between goods and the results yielded by the analysis of the transition matrices agrees with the founded heterogeneity.

Fourth, the study contributes evidence to understand the origin of this heterogeneity. Two alternative strategies are used. The first is based on a set of variables collected in the survey. The second is based on the use of an experimental design. In the first strategy, different determinants are explored, inquiring about the empirical relevance of demographic variables (sex, age), socioeconomic variables (educational level of the respondent, educational level of the parents, number of household members, whether the young person lives independently of their parents) and a set of measures that approximate the position-

ality of the parents (questions that reveal the relevance of interpersonal comparisons of income and the visibility of the goods). The level of positionality is positively associated with young people who no longer live with their parents, who have low levels of education, and whose parents have high levels of education. No association is found between young people's positionality parameters and parents' responses on the relevance of interpersonal comparisons and the visibility of goods. These results do not allow us to reject the intergenerational transmission of the positionality parameter, as an attenuation bias could explain it due to measurement errors. On the other hand, the magnitude of the coefficients and the weak significance of the variables used could indicate that the origin of positionality is associated with institutional factors that are not being considered in the models (Hopkins (2008)' hypothesis). On the other hand, it could also respond to unobservable individual characteristics, such as personality and skills.

The heterogeneity found in the levels of positionality and its connection to inequality aversion introduces new insights into these parameters. First, unequal access to consumption does not affect the well-being of all young people equally, leading to disparities in behavior driven by positional motivations. The findings indicate that individuals from more advantageous socioeconomic backgrounds with lower education levels may have a stronger positional concern. However, this aspect requires further exploration.

Finally, our results suggest that positional concern (driven by inequality aversion) is significantly lower when inequality is explained as a result of effort and inheritance. Participants are more tolerant of inequality resulting from "inheritance" and, to a lesser extent, from "effort," as both seem to be perceived as individual merits. It is pertinent to note that the positionality measures refer to the preferred situation for the (imaginary) grandchild of the participants, so it could be interpreted that there is a direct link between the savings resulting from the participant's effort and the inheritance received by the following generations. This could explain why inheritance is interpreted as the result of effort and therefore is perceived as merit that justifies greater tolerance for inequality. This leads respondents to have a greater tolerance for income inequality. These elements could be relevant for the design of distributive policies and taxes.

References

- Abel, A. B. (1990). Asset Prices Under Habit Formation And Catching Up With The Joneses. *The American Economic Review*, 80(2):38–42. 7
- Agarwal, S., Qian, W., and Zou, X. (2021). Thy Neighbor's Misfortune: Peer Effect on Consumption. *American Economic Journal: Economic Policy*, 13(2):1–25. 4
- Akerlof, G. A. and Kranton, R. E. (2000). Economics and identity. *Quarterly Journal of Economics*, 115(3):715–753. 32
- Almås, I., Cappelen, A. W., and Tungodden, B. (2020). Cutthroat capitalism versus cuddly socialism: Are americans more meritocratic and efficiency-seeking than Scandinavians? *Journal of Political Economy*, 128(5):1753–1788. 4
- Alpizar, F., Carlsson, F., and Johansson-Stenman, O. (2005). How much do we care about absolute versus

relative income and consumption? *Journal of Economic Behavior and Organization*, 56(3):405–421. 3, 4, 5, 7, 9, 12, 13, 20, 24, 25, 29, 34, 52

- Alves, G., Leites, M., and Salas, G. (2022). See it to believe it. Experimental evidence on status good consumption among the youth. *DT IECON 12/22*. 5
- Amiel, Y., Bernasconi, M., Cowell, F., and Dardanoni, V. (2015). Do we value mobility? Social Choice and Welfare, 44(2):231–255. 3, 8
- Amiel, Y. and Cowell, F. A. (1992). Measurement of income inequality. Experimental test by questionnaire. *Journal of Public Economics*, 47(1). 3
- Bastani, S. and Waldenström, D. (2021). Perceptions of Inherited Wealth and the Support for Inheritance Taxation. *Economica*, 88(350):532–569. 5
- Bauer, M., Chytilová, J., and Miguel, E. (2020). Using survey questions to measure preferences: Lessons from an experimental validation in Kenya. *European Economic Review*, 127:103493. 9
- Ben-Ner, A., Kramer, A., and Levy, O. (2008). Economic and hypothetical dictator game experiments: Incentive effects at the individual level. *Journal of Socio-Economics*, 37(5):1775–1784. 9
- Bergolo, M., Burdin, G., Burone, S., de Rosa, M., Giaccobasso, M., and Leites, M. (2022). Dissecting inequality-averse preferences. *Journal of Economic Behavior & Organization*, 200:782–802. 3, 4, 13, 16, 19
- Bertrand, M. (2011). New perspectives on gender. In Card, D. and Ashenfelte, O., editors, *Handbook of Labor Economics*, volume 4B, chapter 17, pages 1543–1590. Elsevier. 5, 28
- Bertrand, M. and Morse, A. (2016). Trickle-down consumption. *Review of Economics and Statistics*, 98(5):863–879. 4
- Beshears, J., Choi, J. J., Laibson, D., and Madrian, B. C. (2008). How are preferences revealed? *Journal of Public Economics*, 92(8-9):1787–1794. 3, 8, 32
- Bisin, A. and Verdier, T. (2011). The economics of cultural transmission and socialization. In Benhabib, J., Bisin, A., and Jackson, M., editors, *Handbook of social economics*, chapter 9, pages 339–416. North-Holland. 2, 6
- Boskin, M. J. and Sheshinski, E. (1978). Optimal redistributive taxation when individual welfare depends upon relative income. *Quarterly Journal of Economics*, 92(4):589–601. 7
- Burone, S. and Leites, M. (2021). Self-centered and non-self-centered inequality aversion matter: Evidence from Uruguay based on an experimental survey. *Journal of Economic Inequality*, 19(2):265– 291. 7
- Bursztyn, L., Ferman, B., Fiorin, S., Kanz, M., and Rao, G. (2018). Status Goods: Experimental evidence from platinum credit cards. *Quarterly Journal of Economics*, 133(3):1561–1595. 4

- Cappelen, A., List, J., Samek, A., and Tungodden, B. (2020). The Effect of Early Education on Social Preferences. *Journal of Political Economy*, 128(7):2739–2758. 4
- Carlsson, F., Daruvala, D., and Jaldell, H. (2010). Value of statistical life and cause of accident: A choice experiment. *Risk Analysis: An International Journal*, 30(6):975–986. 4, 5
- Carlsson, F., Daruvala, D., and Johansson-Stenman, O. (2005). Are people inequality-averse, or just risk-averse? *Economica*, 72(287):375–396. 4, 11
- Carlsson, F., Gupta, G., and Johansson-Stenman, O. (2009). Keeping up with the Vaishyas? Caste and relative standing in India. *Oxford Economic Papers*, 61(1):52–73. 4, 5
- Carlsson, F., Nam, P. K., Linde-Rahr, M., and Martinsson, P. (2007). Are Vietnamese farmers concerned with their relative position in society? *Journal of Development Studies*, 43(7):1177–1188. 4, 5
- Chai, A., Kaus, W., and Kiedaisch, C. (2019). Conspicuous spending and the income distribution of social groups. *Economic Inquiry*, 57(3):1324–1341. 4
- Chao, A. and Schor, J. B. (1998). Empirical tests of status consumption: Evidence from women's cosmetics. *Journal of Economic Psychology*, 19(1):107–131. 5
- Charles, K. K., Hurst, E., and Roussanov, N. (2009). Conspicuous consumption and race. *Quarterly Journal of Economics*, 124(2):425–467. 4, 5
- Clark, A. and D'Ambrosio, C. (2015). Attitudes to income inequality: Experimental and survey evidence. In *Handbook of Income Distribution. Vol. 2*, pages 1147–1208. Elsevier. 2, 4, 6, 7
- Clark, A., Frijters, P., and Shields, M. A. (2008). Relative Income, Happiness, and Utility: An Explanation for the Easterlin Paradox and Other Puzzles. *Journal of Economic Literature*, 46(1):95–144.
 4
- Clark, A. E., Kristensen, N., and Westergård-Nielsen, N. (2009a). Job Satisfaction and Co-Worker Wages: Status or Signal? *The Economic Journal*, 119(536):430–447. 7
- Clark, A. E. and Oswald, A. J. (1996). Satisfaction and comparison income. *Journal of Public Economics*, 61(3):359–381. 7
- Clark, A. E. and Oswald, A. J. (1998). Comparison-concave utility and following behaviour in social and economic settings. *Journal of Public Economics*, 70(1):133–155. 7
- Clark, A. E., Westergård-Nielsen, N., and Kristensen, N. (2009b). Economic satisfaction and income rank in small neighbourhoods. *Journal of the European Economic Association*, 7(2-3):519–527. 7
- Cullen, Z. and Perez-Truglia, R. (2022). How Much Does Your Boss Make? The Effects of Salary Comparisons. *Journal of Political Economy*, 130(3):766–822. 4
- Dannenberg, A., Johansson-Stenman, O., and Wetzel, H. (2022). Status for the good guys: An experiment on charitable giving. *Economic Inquiry*, 60(2):721–740. 4

- De Giorgi, G., Frederiksen, A., and Pistaferri, L. (2020). Consumption Network Effects. *The Review of Economic Studies*, 87(1):130–163. 4
- Duesenberry, J. S. (1967). *Income, saving, and the theory of consumer behavior*. Oxford University Press. 2, 6, 7
- Fehr, E. and Schmidt, K. M. (1999). A theory of fairness, competition, and cooperation. *Quarterly Journal of Economics*, 114(3):817–868. 7
- Fehr, E. and Schmidt, K. M. (2003). Theories of fairness and reciprocity: Evidence and economic applications. In Dewatripont, M., Hansen, L. P., and Turnovsky, S. J., editors, *Advances in Economics and Econometrics: Theory and Applications. Eighth World Congress*, volume 1. Cambridge University Press. 2, 4, 6, 7
- Ferrer-i Carbonell, A. (2005). Income and well-being: An empirical analysis of the comparison income effect. *Journal of Public Economics*, 89(5-6):997–1019. 4
- Fisman, R., Gladstone, K., Kuziemko, I., and Naidu, S. (2020). Do Americans want to tax wealth? Evidence from online surveys. *Journal of Public Economics*, 188:104207. 5
- Frank, R. H. (1985). The demand for unobservable and other nonpositional goods. *American Economic Review*, 75(1):101–116. 2, 6, 7, 8
- Frank, R. H., Posner, R., and Mullainathan, S. (2005). Positional externalities cause large and preventable welfare losses. *American Economic Review*, 95(2):137–141. 2, 6
- Gaertner, W. and Schokkaert, E. (2012). *Empirical social choice: Questionnaire-experimental studies* on distributive justice. Cambridge University Press. 9, 32
- Heffetz, O. (2011). A test of conspicuous consumption: Visibility and income elasticities. *Review of Economics and Statistics*, 93(4):1101–1117. 5, 13, 31, 46
- Heffetz, O. (2012). Who sees what? Demographics and the visibility of consumer expenditures. *Journal* of *Economic Psychology*, 33(4):801–818. 5
- Heffetz, O. (2018). Expenditure Visibility and Consumer Behavior: New Evidence. National Bureau of Economic Research, w25161. 5
- Heffetz, O. and Frank, R. H. (2011). Preferences for status: Evidence and economic implications. In *Handbook of Social Economics*, volume 1, pages 69–91. North-Holland. 2, 7
- Hirsch, F. (1976). Social limits to growth. Cambridge, Mass: Harvard University Press. 2, 8
- Hirschman, A. and Rothschild, M. (1973). The changing tolerance for income inequality in the course of economic development. *Quarterly Journal of Economics*, 87(4):544–566. 31
- Holbrook, A. L., Green, M. C., and Krosnick, J. A. (2003). Telephone versus face-to-face interviewing of national probability samples with long questionnaires: Comparisons of respondent satisficing and social desirability response bias. *Public Opinion Quarterly*, 67(1):79–125. 8

- Hopkins, E. (2008). Inequality, happiness and relative concerns: What actually is their relationship? *Journal of Economic Inequality*, 6(4):351–372. 2, 6, 7, 36
- Johansson-Stenman, O., Carlsson, F., and Daruvala, D. (2002). Measuring future grandparents' preferences for equality and relative standing. *Economic Journal*, 112(479):362–383. 7, 9, 11, 29
- Kahneman, D. and Knetsch, J. L. (1992). Valuing public goods: The purchase of moral satisfaction. *Journal of Environmental Economics and Management*, 22(1):57–70. 32
- Kaus, W. (2013). Conspicuous consumption and "race": Evidence from South Africa. *Journal of Development Economics*, 100(1):63–73. 4, 5
- Khamis, M., Prakash, N., and Siddique, Z. (2012). Consumption and social identity: Evidence from India. *Journal of Economic Behavior and Organization*, 83(3):353–371. 5
- Kuhn, P., Kooreman, P., Soetevent, A., and Kapteyn, A. (2011). The effects of lottery prizes on winners and their neighbors: Evidence from the Dutch postcode lottery. *American Economic Review*, 101(5):2226–2247. 4
- Larney, A., Rotella, A., and Barclay, P. (2019). Stake size effects in ultimatum game and dictator game offers: A meta-analysis. *Organizational Behavior and Human Decision Processes*, 151:61–72. 9
- Layard, R. (1980). Human Satisfactions and Public Policy. The Economic Journal, 90(360):737–750. 7
- Leites, M., Paleo, C., Ramos, X., and Salas, G. (2022). Choosing or Inheriting the Joneses: The origins of reference groups. *DT IECON*, 04/22. 31
- Leites, M., Salas, G., and Vigorito, A. (2019). Bienes visibles, rasgos de personalidad y preferencias por estatus. *DT IECON*, 19-19. 13, 31, 46
- Perez-Truglia, R. (2020). The effects of income transparency on well-being: Evidence from a natural experiment. *American Economic Review*, 110(4):1019–1054. 4
- Postlewaite, A. (1998). The social basis of interdependent preferences. *European Economic Review*, 42(3-5):779–800. 2, 6
- Robson, A. J. (1992). Status, the Distribution of Wealth, Private and Social Attitudes to Risk. *Econometrica*, 60(4):837–857. 7
- van Praag, B. (2011). Well-being inequality and reference groups: An agenda for new research. *Journal* of *Economic Inequality*, 9(1):111–127. 7
- Veblen, T. (1994). The Theory of the Leisure Class. 1899. Mineola, NY: Dover. 2, 6, 31
- Walker, D. M. and Frank, R. H. (1999). Luxury Fever: Why Money Fails to Satisfy in an Era of Excess. *Southern Economic Journal*, 66(1):199–201. 28
- Weiss, Y. and Fershtman, C. (1998). Social status and economic performance: A survey. *European Economic Review*, 42(3-5):801–820. 2

Appendix A Appendix of Tables and Figures

Tables

Table A1: Societies in the relative-income experiment and the parameters implied Marginal degree of positionality

Society	Grandchild Income	Mean income	γ indifference
А	37,500	45,000	
C_1	37,500	30,000	0
$\dot{C_2}$	35,625	30,000	0.125
$\overline{C_3}$	30,625	30,000	0.5
C_4	27,500	30,000	0.75
C_5	25,000	30,000	1

The amounts of money are expressed in Uruguayan pesos (UYU). 1 USD = 32 UYU.

Table A2: Societies A and	C_z in	the relative	consumption	experiment
---------------------------	----------	--------------	-------------	------------

Societies Societies	Medical Insurance (in UYU)		Medical InsuranceJewelry(in UYU)(in USD)		Incor (in UY	ne (U)
	Grandchild	Average	Grandchild	Average	Grandchild	Average
А	7,500	9,000	1,000	1,200	37,500	45,000
C_1	7,500	6,000	1,000	800	37,500	30,000
C_2	7,125	6,000	950	800	35,625	30,000
$\tilde{C_3}$	6,125	6,000	817	800	30,625	30,000
C_4	5,500	6,000	733	800	27,500	30,000
C_5	5,000	6,000	667	800	25,000	30,000

In the columns referring to medical insurance and income, the amounts of money are expressed in Uruguayan pesos (UYU). In the jewelry column, the money amounts are expressed in dollars (USD). 1 USD = 32 UYU.

Table A3: Samples sizes according to the alternatives criteria to define consistent responses

Society	γ: Break point	Set of choices (Baseline Sample)	Alternative set of choices (Extended Sample)
A	less than 0	ΑΑΑΑΑ	AAAAA AAAAC ₅ AC ₂ AAA AAC ₃ AA AAAC ₄ A
$\begin{array}{c} C_1\\ C_2\\ C_3\\ C_4 \end{array}$	0 - 0.125 0.125 - 0.5 0.5 - 0.75 0.75 - 1	$C_1 AAAA C_1 C_2 AAA C_1 C_2 C_3 AA C_1 C_2 C_3 C_4 A$	
<i>C</i> ₅	more than 1	$C_1 C_2 C_3 C_4 C_5$	$\begin{array}{c} C_{1}C_{2}C_{3}C_{4}C_{5} \\ AC_{2}C_{3}C_{4}C_{5} \\ AAC_{4}C_{4}C_{5} \\ C_{1}C_{2}C_{3}AC_{5} \\ C_{1}AC_{3}C_{4}C_{5} \\ C_{1}C_{2}AC_{4}C_{5} \end{array}$

Table A4: Mean of demographic and socioeconomic variables. Different samples and consistency criteria. Sample A

		Group	ρI			p-value	
	Jewelry (1)	Insurance (2)	Income (3)	Car (4)	(1) vs. (4)	(2) vs. (4)	(3) vs. (4)
I. Baseline Criterion							
Age	20.24	20.19	20.13	20.16	0.144	0.669	0.589
Male	0.513	0.480	0.488	0.464	0.284	0.717	0.568
Emancipated	0.152	0.176	0.152	0.168	0.638	0.813	0.603
Household size	4.043	3.769	3.878	3.908	0.415	0.357	0.836
Work (1=Yes)	0.384	0.392	0.392	0.388	0.934	0.927	0.927
< 10 years of education	0.358	0.330	0.288	0.300	0.177	0.467	0.751
II. Extended Criterion							
Age	20.19	20.17	20.16	20.16	0.452	0.707	0.974
Male	0.486	0.464	0.487	0.476	0.773	0.721	0.746
Emancipated	0.153	0.146	0.160	0.150	0.908	0.875	0.675
Household size	3.944	3.922	3.934	3.906	0.758	0.980	0.811
Work (1=Yes)	0.367	0.380	0.400	0.382	0.676	0.954	0.580
< 10 years of education	0.327	0.321	0.321	0.300	0.409	0.507	0.497

		Baseline C	Criterion			Extended (Criterion	
	Random	Effect	Int. R	eg.	Random	Effect	Int. R	.eg.
Health Insurance	0.205***	0.221***	0.527***	0.770***	0.266***	0.282***	1.003***	1.138**
	(0.032)	(0.056)	(0.084)	(0.215)	(0.027)	(0.034)	(0.105)	(0.152)
Jewelry	0.205***	0.298***	0.542***	1.133***	0.247***	0.283***	0.947***	1.166**
	(0.035)	(0.055)	(0.091)	(0.224)	(0.029)	(0.035)	(0.109)	(0.153)
Car	0.222*** (0.038)		0.549*** (0.108)		0.252*** (0.032)		0.912*** (0.130)	
Age	-0.046	0.021	-0.156	0.142	-0.038*	-0.015	-0.244	-0.238
	(0.029)	(0.198)	(0.124)	(0.729)	(0.022)	(0.029)	(0.166)	(0.241)
Age2	0.001*	-0.000	0.003	-0.003	0.001**	0.000	0.005	0.005
	(0.000)	(0.004)	(0.002)	(0.016)	(0.000)	(0.000)	(0.003)	(0.005)
Female	0.034	0.004	0.099	-0.010	0.051**	0.045	0.214**	0.191
	(0.031)	(0.060)	(0.079)	(0.235)	(0.026)	(0.035)	(0.100)	(0.151)
Household size	-0.010	-0.008	-0.014	-0.012	-0.010	-0.008	-0.025	-0.013
	(0.010)	(0.018)	(0.023)	(0.072)	(0.007)	(0.010)	(0.029)	(0.044)
Emancipated	0.062	0.186**	0.184	0.835***	0.011	0.044	0.059	0.222
	(0.045)	(0.079)	(0.114)	(0.323)	(0.038)	(0.047)	(0.143)	(0.209)
< 10 years of education	-0.004	0.027	0.016	0.091	-0.009	-0.021	0.002	-0.107
	(0.035)	(0.064)	(0.088)	(0.249)	(0.030)	(0.040)	(0.111)	(0.165)
Work (omitted: < 30 hou	rs)							
30-40 hours	0.069*	0.045	0.180*	0.143	0.062**	0.062	0.245**	0.261
	(0.036)	(0.075)	(0.096)	(0.286)	(0.031)	(0.042)	(0.119)	(0.182)
>40 hours	0.044	0.012	0.103	-0.012	0.030	0.029	0.093	0.111
	(0.036)	(0.062)	(0.093)	(0.267)	(0.030)	(0.041)	(0.118)	(0.177)
Sample	0.022	0.044	0.061	0.143	-0.000	0.032	0.011	0.093
	(0.032)	(0.061)	(0.083)	(0.236)	(0.027)	(0.035)	(0.105)	(0.153)
Constant	1.167***	0.214	2.328	-1.562	1.113***	0.730*	3.444*	3.078
	(0.404)	(2.207)	(1.519)	(8.315)	(0.318)	(0.425)	(2.017)	(2.928)
σ_u			0.556*** (0.056)	0.941*** (0.156)			0.812*** (0.069)	0.882** (0.109)
σ_e			1.102*** (0.047)	1.683*** (0.130)			1.548*** (0.058)	1.764** (0.090)
Obs.	1638	561	1638	561	2436	1350	2436	1350

Table A5: Multivariate regression estimates for γ - Full estimation

*Significant at 10%, **significant at 5%, ***significant at 1%. Standard errors are in parentheses.

Table A6: Balance checks by treatment arm: grandchild vs. granddaughter groups. Sample A

		All		Con	sistent with Incom	e
	Grandson	Granddaughter	t test	Grandson	Granddaughter	t test
Age	20.14	20.16	0.316	20.11	20.17	0.925
Male	0.424	0.508	1.968	0.472	0.497	0.489
Emancipated	0.149	0.147	0.035	0.142	0.166	0.624
Household size	4.029	3.867	1.079	3.875	3.872	0.019
Work (1=Yes)	0.435	0.347	2.131	0.438	0.349	1.717
At high-school (1=Yes)	0.520	0.478	0.982	0.561	0.511	0.947
< 10 years of education	0.320	0.320	0.004	0.295	0.289	0.140
N	276	278		176	187	

		I. All		II. Inco	me-treatments c	onsistency
	Effort	Inheritance	t test	Effort	Inheritance	t test
Age	20.82	20.75	0.236	20.93	20.76	0.382
Male	0.482	0.500	0.364	0.484	0.527	0.611
Emancipated	0.168	0.194	0.687	0.227	0.218	0.148
Household size	3.882	3.813	0.424	3.684	3.972	1.194
Work (1=Yes)	0.429	0.413	0.336	0.454	0.427	0.379
At high-school (1=Yes)	0.597	0.612	0.300	0.588	0.600	0.180
< 10 years of education	0.236	0.277	0.935	0.237	0.282	0.728
N	191	206		97	110	

Table A7: Balance checks by treatment arm: effort vs. inheritance groups. Sample B

The balance checks are made for all Sample B (panel L) and the subset of those responses consistent in the income scenario (γ^{income}), considering the controls and those treated by inheritance and effort (Income-treatments consistency) (panel II).

Table A8: Marginal degree of positionality of health insurance ($\gamma^{insurance}$) by public health insurance assessment, the relevance of health in the live, and employment status

	I.	Baselin	e Criterion		II.	Extende	ed Criterion	
	Group	I	Group	II	Group	I	Group	Π
	$\gamma^{\text{insurance}}$	N	$\gamma^{\text{insurance}}$	N	$\gamma^{\text{insurance}}$	N	$\gamma^{\text{insurance}}$	N
A. Assessment of public health	h insurance (Sample	A [†])					
(1) Very bad, bad or regular	0.81	42	0.72	16	0.91	61	0.85	33
(2) Good or very good	0.87	117	0.80	47	0.96	175	0.98	120
I do not known	0.98	21	0.80	8	1.00	42	0.89	24
Mean test - (1) vs. (2)								
p-value	0.54		0.61		0.47		0.19	
B. Labour market status (Sam	ple A and B)							
(1) Does not work	0.86	277	0.80	112	0.95	441	0.94	283
(2) Work	0.77	200	0.69	83	0.87	297	0.86	191
Mean test - (1) vs. (2)								
p-value	0.06		0.20		0.06		0.11	
C. How important is health an	d nutrition in	your li	fe (Sample A	(†)				
(1) Low priority (score: 1-6)	0.79	34	0,65	12	0,89	54	0.93	28
(2) High priority (score: 7-9)	0.88	157	0.81	65	0.97	246	0.95	163
Mean test - (1) vs. (2)								
p-value	0.32		0.36		0.27		0.82	

[†] The information about respondents' assessment of public health insurance and the priority to health in their lives was elicited in wave 4 of ELBU, 3-4 years before conducting the consumer experiment (Sample A lagging). We do not have this information for the friends of the ELBU youth (Sample B). The results are only tentative due to the number of observations (particularly concerning the priority of health in the respondents' lives).

	Bas	seline Criterio	on	Ext	ended Criteri	on
Good* Good [†]	Jewerly Insurance	Car Insurance	Car Jewerly	Jewerly Insurance	Car Insurance	Car Jewerly
Transition matrix						
% persistence $(\gamma^{\text{good}^{\star}} = \gamma^{\text{good}^{\dagger}})$	0.57	0.31	0.31	0.60	0.36	0.34
$\% \gamma^{\text{good}^{\star}} \geq \gamma^{\text{good}^{\dagger}}$	0.80	0.77	0.73	0.79	0.83	0.78
Conditional distribution test: p - value						
$H_0: \gamma_i^{\text{good}\star} < \gamma_i^{\text{good}\star} \text{ if } \gamma_i^{\text{good}\dagger} \le 0.5 \text{ and } \gamma_i^{\text{good}\dagger} > 0.5$	0.01	0.14	0.01	0.00	0.09	0.00
$H_0: \gamma_i^{\text{good}\star} > \gamma_i^{\text{good}\star} \text{ if } \gamma_i^{\text{good}\dagger} \le 0.5 \text{ and } \gamma_i^{\text{good}\dagger} > 0.5$	1.00	1.00	1.00	1.00	1.00	1.00
$H_0: F_i(\gamma^{\text{good}\star} \mid \gamma^{\text{good}\dagger} \le 0.5) = F_i(\gamma^{\text{good}\star} \mid \gamma^{\text{good}\dagger} > 0.5)$	0.00	0.25	0.01	0.00	0.16	0.00
Ν	237	353	297	540	639	567

Table A9: Synthetic index and conditional distribution tests. Conditional variable: health insurance or jewelry. Entire sample

This Table displays the synthetic measures from the transition matrix and the test of differences on the conditional distributions $F_i(\gamma^{\text{goods}*} | \gamma_a^{\text{good}} \leq 0.5)$ and $F_i(\gamma^{\text{good}*} | \gamma_b^{\text{good}}) > 0.5$. We report the frequency of persistence between $\gamma^{\text{good}*}$ and $\gamma^{\text{good}*}$ (the sum of the unconditioned frequencies of the matrix's main diagonal). p-values Kolmogorov-Smirnov tests for the three alternative hypotheses. The p-value of the equal distribution test is estimated using the exact option.

 Table A10: Effort/Inheritance. Different Specifications and consistency criteria. Sample B

	Randor	n Effect	Fixed	Effect	Int.	Reg.
I. Baseline Criterion						
Omitted variable: Only income -	control group	р-				
Effort/Inheritance	-0.065**	-0.054*	-0.054*	-0.054*	-0.113**	-0.102*
	(0.030)	(0.032)	(0.032)	(0.032)	(0.054)	(0.060)
Constant	0.058	0.014	0.622***	0.593***	-0.709	-0.459
	(0.484)	(0.594)	(0.010)	(0.016)	(1.014)	(1.397)
Observations	841	404	841	404	841	404
R2	0.014	0.014	0.014	0.014		
II. Extended Criterion						
Omitted variable: Only income -	control group	р-				
Effort/Inheritance	-0.074**	-0.071**	-0.071**	-0.071**	-0.143**	-0.137**
	(0.030)	(0.032)	(0.031)	(0.031)	(0.063)	(0.065)
Constant	-0.041	-0.114	0.652***	0.651***	-0.888	-0.784
	(0.465)	(0.549)	(0.009)	(0.016)	(1.185)	(1.379)
Observations	1,010	502	1,010	502	1,010	502
R2	0.020	0.020	0.020	0.020		
Controls	Yes	Yes	No	No	Yes	Yes
Income-treatments consistency	No	Yes	No	Yes	No	Yes

In this case, the estimates are made for all Sample B and the subset of those responses consistent in the income scenario (γ^{income}), considering the controls and those treated by inheritance and effort (Income-treatments consistency). The covariates included as controls are: age; age squared; sex (1=female); household size; if the young person left the home of origin (1=Yes); years of education (1=<10 years of education); and labor condition based on two dummy variables (work between 30 and 40 hours a week, and work more than 40 hours a week). *Significant at 10%, **significant at 5%, ***significant at 1%. Standard errors are in parentheses.

		Sample	e A			Sample	в	
_	Mean	Std. Dev.	Min.	Max.	Mean	Std. Dev.	Min.	Max.
I Comparison with othe	r naonla E	ntira Sampla						
	i people. E		1.00	10.00	0.20	2.09	1.00	10.00
Looking up	5.94	3.52	1.00	10.00	8.20	2.08	1.00	10.00
Looking down	5.87	3.51	1.00	10.00	8.15	2.02	1.00	10.00
II. Visibility of good. O	nly Sample	А						
Visibility Index (ln)	-0.35	0.63	-3.00	0.00				
III. Consumption of jew	elry (watch	. chain. ring)	. Only Sa	mple B				
Successful people		,,	· · · · · · · · · · · · · · · · · · ·		0.03	0.26	0.00	1.00
Succession people					0.95	0.20	0.00	1.00
Improves social position					0.66	0.47	0.00	1.00

Table A11: Preference for status. Extended Criterion

In the Looking up, the question is: "Imagine that you get an offer of a permanent full-time job that you like. Your potential employer asks you to indicate the wage you are willing to receive. For each of the following items, please indicate on a scale from 1 to 10 (where 1 is very little and 10 is a lot) how true it is that you would consider, in your proposal, the wage of [reference group]", where the possible reference groups are: friends, family, neighbors, people with the same job profile, and union workers. Respondents are asked in the Looking down: "Imagine an economic crisis, and your household income is reduced. Indicate on a scale from 1 to 10 (where 1 is very little and 10 is a lot) how true it is that your economic satisfaction would be affected if your income falls below the income of [reference group]", where possible reference groups are the same as in the question Looking up. We compute the maximum value of each respondent's answers across all external reference groups separately for the Looking up and Looking down scenarios. The Visible Index is based on the following question: "Imagine that you meet a new person who lives in a household similar to yours. Imagine that their household is not different from other similar households except that they like to, and do, consistently spend more than average on (category of good). How closely would you have to interact with them to observe the consistently above-average spending (on each category of goods)? Would you notice it almost immediately upon meeting them for the first time, a short while after, a while after, only a long while after, or never?" This question was repeated for 31 categories of goods. As Heffetz (2011), we first recorded the responses as follows: zero (answer=3: no visibility), 0.5 (answer =2: medium visibility), and 1 (answer =1: maximum visibility). Then for each individual, we calculated the average value of the 10 (out of 31 possible) items in Leites et al. (2019) to show the highest level of visibility for young people in Uruguay. Finally, the Successful People variable is built from the question, "Imagine a person who is successful in what they do. Do they have any of the following goods?" The variable Improves Social Position arises from the question, "Do you think that if a person increases their consumption/use of one of the following goods, their social position improves according to the opinion of the rest of society?" In both cases, fifteen different goods are listed. A dummy variable is constructed if the interviewee answers that if in the ring, watch, or chain options, the rest include goods such as sunglasses, leather jackets, or signature tattoos

		Jewelry		Ca	ы		Insurance			Income	
	(E)	(2)	(3)	(4)	(2)	(9)	6	(8)	(6)	(10)	(11)
Age	-0.100 (0.154)	-5.055 (5.202)	-0.217 (0.179)	-3.616 (4.945)	-4.659 (4.995)	-0.208 (0.150)	2.204 (6.906)	-0.313** (0.151)	0.039 (0.147)	-3.576 (5.495)	-0.126 (0.163)
Age×Age	0.002 (0.002)	0.126 (0.127)	0.004 (0.003)	0.086 (0.122)	0.111 (0.123)	0.003 (0.002)	-0.055 (0.169)	0.005** (0.002)	-0.000 (0.002)	0.091 (0.135)	0.003 (0.002)
Female	0.318* (0.175)	0.577** (0.227)	-0.043 (0.284)	0.204 (0.223)	0.224 (0.222)	0.112 (0.159)	0.206 (0.211)	0.045 (0.245)	0.223 (0.160)	0.124 (0.213)	0.417* (0.251)
Household size	-0.105** (0.052)	-0.091 (0.066)	-0.138* (0.082)	-0.033 (0.074)	-0.023 (0.075)	0.010 (0.040)	0.035 (0.049)	-0.028 (0.067)	-0.043 (0.044)	-0.056 (0.056)	0.002 (0.072)
Emancipated	0.140 (0.238)	0.011 (0.310)	0.192 (0.380)	0.021 (0.334)	0.018 (0.332)	0.121 (0.221)	0.021 (0.299)	0.198 (0.353)	0.022 (0.238)	-0.072 (0.325)	0.166 (0.348)
< 10 years of education	0.213 (0.188)	0.268 (0.243)	-0.046 (0.328)	0.257 (0.254)	0.243 (0.258)	-0.220 (0.185)	0.008 (0.238)	-0.758** (0.305)	-0.038 (0.186)	0.136 (0.249)	-0.489* (0.290)
Work (omitted: < 30 hours)											
30-40 hours)	0.025 (0.208)	0.060 (0.274)	0.063 (0.327)	0.168 (0.258)	0.173 (0.258)	0.559*** (0.179)	0.301 (0.249)	0.865*** (0.256)	0.123 (0.188)	0.333 (0.257)	-0.121 (0.286)
>40 hours	0.088 (0.208)	0.168 (0.271)	-0.070 (0.344)	0.074 (0.266)	0.102 (0.267)	0.094 (0.199)	0.003 (0.251)	0.252 (0.325)	0.102 (0.192)	-0.122 (0.252)	0.559* (0.296)
Sample	0.368* (0.206)					0.234 (0.191)			-0.235 (0.184)		
Comparison with other people											
Looking up	0.118*** (0.045)			0.038 (0.054)		0.046 (0.041)			0.112*** (0.038)		
Looking down	-0.083* (0.045)			-0.010 (0.058)		-0.053 (0.043)			-0.137*** (0.038)		
Visibility Index (ln)		0.318 (0.212)			-0.025 (0.196)		-0.021 (0.187)			-0.355** (0.175)	
Consumption of jewelry (watch	ı, chain, ring)										
Successful people			-0.721* (0.436)					-0.820*** (0.296)			-0.426 (0.446)
Improves social position			-0.184 (0.301)					-0.627*** (0.241)			-0.098 (0.250)
Constant	5.537** (2.158)	55.635 (53.139)	8.530*** (2.565)	41.986 (50.216)	53.048 (50.656)	7.412*** (2.161)	-17.841 (70.692)	10.159*** (2.230)	3.283 (2.090)	38.406 (55.818)	5.462** (2.388)
Obs. R2	623 0.038	365 0.039	258 0.029	382 0.017	382 0.010	699 0.033	392 0.009	307 0.096	732 0.027	414 0.021	318 0.033

*Significant at 10%, **significant at 5%, ***significant at 1%. Standard errors in parentheses.

- EC. Full estimation
for γ
ates
stim
regression e
'ariate
Multiv
A12:
Table

	I.Baseli	ne Criterion Wanted num	II. Exter	nded Criterion dren
	0	>0	0	>0
Group II - % consistent	18.6	22.0	48.1	51.2
Group I - % consistent w	ith			
Health insurance	50.8	52.8	75.0	84.3
Jewelry	39.6	43.2	71.9	69.6
Car	45.6	55.2	76.2	82.4
Income	65.9	69.6	82.6	84.1
N	260	127	260	127

Table A13: Analysis of the consistency of responses by the desired number of children of respondents

Consistent rate based on people who participate in ELBU (sample A). The information about the wanted number of children is from wave 4 of the ELBU.

Table A14: Average marginal degree of positionality by desired number of children

	I. E	Baseline	Criterio	n	П.	Extend	led Criter	ion
	Gro	up I	Grou	p II	Gro	up I	Gro	up II
	γ	N	γ	N	γ	N	γ	Ν
I. The respon	dent dec	clares th	at he/sh	e does	not wis	h to hav	e childre	n
Health Insurance	0.82	132	0.79	49	0.93	195	0.94	127
Jewelry	0.86	103	0.89	49	0.93	187	0.93	127
Income	0.63	174	0.59	49	0.65	218	0.62	127
II. The respon	ndent de	clares the	hat he/sh	ne does	s not wis	h to ha	ve childre	en
Health Insurance	0.95	60	0.79	28	1.00	107	0.95	65
Jewelry	0.80	54	0.90	28	0.93	87	0.94	65
Income	0.60	86	0.45	28	0.57	106	0.61	65
Test of difference b	y the nu	umber o	f desired	l child	ren of th	e respo	ndents - j	p-value
Health insurance	0.11		0.99		0.22		0.95	
Jewerly	0.51		0.94		0.92		0.90	
Income	0.71		0.29		0.22		0.91	

	Ba	Baseline Criterion				Extended Criterion			
	Group I		Group II		Group I		Group II		
	γ	Ν	γ	Ν	γ	N	γ	N	
L Risk averse									
Jewelry	4.35	46	3.92	26	4.66	68	4.57	51	
Insurance	3.82	60	3.85	26	4.22	79	4.22	51	
Income	3.46	65	3.23	26	3.72	79	3.49	51	
II. Risk lovers									
Jewelry	4.49	43	4.05	20	4.62	69	4.47	51	
Insurance	4.18	51	3.65	20	4.63	90	4.37	51	
Income	4.00	72	3.65	20	4.10	94	3.88	51	
III. Risk neutral									
Jewelry	4.27	60	4.38	34	4.51	117	4.72	86	
Insurance	4.45	85	4.35	34	4.87	134	4.94	86	
Income	3.53	114	3.21	34	3.61	143	3.41	86	
Test of difference by ri Jewerly	sk - p-va	alue							
Averse vs. Neutral	0.85		0.45		0.65		0.68		
Averse vs. Lover	0.76		0.85		0.91		0.82		
Lover vs. Neutral	0.61		0.61		0.74		0.51		
Insurance									
Averse vs. Neutral	0.09		0.41		0.03		0.05		
Averse vs. Lover	0.41		0.78		0.22		0.73		
Lover vs. Neutral	0.47		0.28		0.39		0.11		
Income									
Averse vs. Neutral	0.84		0.96		0.70		0.83		
Averse vs. Lover	0.11		0.52		0.24		0.36		
Lover vs. Neutral	0.11		0.48		0.08		0.23		

 Table A15: Average marginal degree of positionality by risk

Table A16: Ave	erage marginal d	legree of positionalit	y. Robustness anal	ysis removing	g extreme values
		<i>()</i>	-		,

		Removing	ŗ
	Lowest range	Highest range	Both extremes
I. Baseline Criterion			
Health Insurance	1.09	0.30	0.65
Jewelry	1.08	0.29	0.62
Car	1.08	0.38	0.72
Income	0.85	0.37	0.59
II. Extended Criterion			
Health Insurance	1.15	0.26	0.65
Jewelry	1.15	0.22	0.62
Car	1.15	0.29	0.72
Income	0.92	0.32	0.59

	Randor	n Effect	Fixed	Effect	Int.	Reg.
I. Baseline Criterion						
Omitted variable: Income						
Health Insurance	0.092***	0.124*	0.069	0.139	0.089**	0.118
	(0.035)	(0.074)	(0.059)	(0.100)	(0.035)	(0.076)
Jewelry	0.044	0.003	-0.083	-0.117	0.053	0.038
	(0.042)	(0.085)	(0.073)	(0.105)	(0.039)	(0.073)
Car [†]	0.125***		0.076		0.123***	
	(0.045)		(0.075)		(0.044)	
Constant	1.232	-0.153	0.605***	0.553***	1.149	-0.326
	(0.822)	(2.327)	(0.023)	(0.042)	(0.854)	(1.989)
Observations	483	107	483	107	483	107
R2	0.023	0.125	0.050	0.174		
II. Extended Criterion						
Omitted variable: Income						
Health Insurance	0.329***	0.403***	0.384***	0.460***	0.677***	0.782***
	(0.032)	(0.044)	(0.041)	(0.052)	(0.072)	(0.098)
Jewelry	0.277***	0.358***	0.304***	0.360***	0.587***	0.706***
	(0.036)	(0.047)	(0.044)	(0.055)	(0.072)	(0.092)
Car [†]	0.281***		0.311***		0.549***	
	(0.041)		(0.048)		(0.085)	
Constant	0.923***	0.483	0.604***	0.563***	2.180	2.472
	(0.318)	(0.421)	(0.024)	(0.031)	(1.616)	(2.418)
Observations	1,281	623	1,281	623	1,281	623
R2	0.152	0.225	0.153	0.228		
Group	I	II	I	II	I	II
Controls	Yes	Yes	No	No	Yes	Yes

Table A17: γ estimation removing both extremes. Different Specifications and consistency criteria

The covariates included as controls are: age; age squared; sex (1=female); household size; if the young person left the home of origin (1=Yes); years of education (1=<10 years of education); and labor condition based on two dummy variables (work between 30 and 40 hours a week, and work more than 40 hours a week). Additionally, we include a dummy variable that identifies whether the young person belongs to sample A or B. *Significant at 10%, **significant at 5%, ***significant at 1%. Standard errors are in parentheses. [†] The hypothetical scenario of the cars was only presented to sample A. Therefore, consistency refers to that population.

Table A18:	Average marginal	degree of	positionality	by order effect
	i) i)			-

	E	Baseline Criterion				Extended Criterion			
	Grou	Group I		Group II		Group I		рII	
	γ	Ν	γ	N	γ	N	γ	N	
Income in 4th position	3.63	260	3.25	83	3.80	329	3.57	195	
Income in 3rd position	3.52	361	3.22	112	3.54	456	3.53	280	
Test of differences of me 4th vs. 3rd position	eans - p-v 0.493	alue	0.924		0.086		0.820		

	I. Baseline	Criterion	II. Extende	d Criterion
	(1)	(2)	(3)	(4)
Omitted variable: Income				
Health Insurance	0.500***	0.750***	0.750***	1.125***
	(0.036)	(0.108)	(0.079)	(0.073)
Jewelry	0.500***	0.750***	0.750***	1.125***
	(0.037)	(0.074)	(0.080)	(0.070)
Car	0.500***		0.750***	
	(0.037)		(0.091)	
Constant	0.750	0.500	0.500	0.125
	(1.720)	(2.000)	(1.179)	(0.770)
Observations	1,631	587	2,432	1350
Pseudo R2	0.082	0.115	0.113	0.144
Group II	No	Yes	No	Yes

Table A19: Quantile regressions estimates analysis for γ . Robustness analysis

The covariates included as controls are: age; age squared; sex (1=female); household size; if the young person left the home of origin (1=Yes); years of education (1=<10 years of education); and labor condition based on two dummy variables (work between 30 and 40 hours a week, and work more than 40 hours a week). Additionally, we include a dummy variable that identifies whether the young person belongs to sample A or B. *Significant at 10%, **significant at 5%, ***significant at 1%. Standard errors are in parentheses. The estimate was made for the 50th quantile.

Figures

Figure A1: Screenshot of baseline instructions

B.4) ...el ingreso mensual de tu nieto/a y el ingreso mensual promedio del resto de las personas de esa sociedad. Además, sabes que el ingreso de tu nieto/a se debe, en gran medida, a una herencia que recibió. Por favor, elegí en cuál de las sociedades que te muestro a continuación, tu nieto/a estará mejor. (MOSTRAR TARIETAS)

		Situación 1	Situa <mark>ción 2</mark>	Situación 3	Situación 4	Situación 5
	Ingreso mensual del nieto/a en 📃 sociedad A	\$ 37500	\$ 37500	\$ 37500	\$ 37500	\$ 37500
Sociedad A	Ingreso medio mensual de la sociedad A	\$ 45000	\$ 45000	\$ 45000	\$ 45000	\$ 45000
Ingreso mensual del nieto/a en sociedad B		\$ 37500	<mark>\$ 3</mark> 5625	\$ 30625	\$ 27500	\$ 25000
Sociedad B	Ingreso medio mensual de la sociedad B	\$ 30000	<mark>\$ 3</mark> 0000 \$ 30000		\$ 30000	\$ 30000
Encuestador I	Para que el encuestado realice las	<mark>elec</mark> ciones, es neo	e <mark>sario que lea</mark> la	s tarjetas corresp	ondientes a cada	i situación,
considerándo	las p <mark>or separado. Es</mark> decir, que pri	<mark>mer</mark> o seleccione s	us preferencias e	en relación a la si	tuación 1, luego a	a la 2 y así
sucesivament	e h <mark>asta</mark> culminar.					
Prefiero socied	ad A					
Prefiero socied	ad B					

Figure A2: Comparative of the marginal degree of positionality γ in Uruguay and Costa Rica



Developed by us using data from the applied questionnaire in Uruguay and data presented by Alpizar et al. (2005) for Costa Rica.





This figure displays the distribution of γ estimated using the Extended Criterion in each case. In the *x-axis*, we report the implied value of γ associated with different choices of *A* and *C_z*. On the *y-axis*, we report the frequency of γ associated with each good and income. The dot-dashed line represents our estimate for the mean using OLS regression of γ over a constant. p-values Kolmogorov-Smirnov tests for equal distribution are presented in each graph.



Figure A4: Matrix transition of subjects a marginal degree of positionality. Baseline Criterion

This figure displays subjects' movements (and their γ^b) when they choose alternative treatments. We created these transition matrices to represent the transition probability between the goods. The rows represent the γ of the first-mentioned goods in the matrix, and the columns the γ of the second-mentioned goods conditional on the other γ included in the matrix. The diagonal of the matrix represents the persistence of subjects in each of the six ranges of γ for the goods considered.



Figure A5: Marginal degree of positionality distribution by gender

This figure displays the distribution of γ estimated using the Baseline Criterion in each case. In the *x*-axis, we report the implied value of γ associated with different choices of *A* and C_z . On the *y*-axis, we report the frequency of γ associated with each good and income. The dot-dashed line represents our estimate for the mean for males and females using OLS regression of γ over a constant. The hypothesis that the means are different is rejected in all cases, even for income and health insurance. The p-value of the mean difference test in income is 0.304 (where the mean value for males is 0.606 and for females is 0.649) and 0.346 in health insurance (where the mean value for males is 0.801 and for females is 0.847). p-values Epps-Singleton tests for equal distribution are presented in each figure.



Figure A6: Marginal degree of positionality distribution by grandchild's gender

This figure displays the distribution of γ estimated using the Baseline Criterion in each case. In the *x-axis*, we report the implied value of γ associated with different choices of *A* and C_z . On the *y-axis*, we report the frequency of γ associated with each good and income. The dot-dashed line represents our estimate for the mean for males and females using OLS regression of γ over a constant. In all cases, the hypothesis that the means are different is rejected, even in the case of income, the p-value of the mean difference test is 0.296 (where the mean value for grandson is 0.587 and granddaughter is 0.643). p-values Epps-Singleton tests for equal distribution are presented in each graph.

Figure A7: Simulation



This figure displays the distribution of γ estimated using in each case the Baseline Criterion (as a benchmark), information based on simulation I, and information based on simulation II. Panel (a) refers to the relative consumption experiment when participants choices the consumption of Health insurance. Panel (b) refers to the relative consumption experiment when participants choices the consumption of jewelry. Panel (c) refers to the relative consumption experiment when participants choices the consumption of cars. The three panels include the results based on the relative income experiments as a benchmark. In the *x*-axis, we report the implied value of γ associated with different choices of A and C_z. On the *y*-axis, we report the frequency of γ associated with each good (bar) and income (dots).

Appendix B A random sample choice between pairs of hypothetical situations

We employed an alternative approach based on hypothetical situations presented in pairs to evaluate the robustness of our findings and potential biases arising from inconsistent responses. Specifically, we randomly selected 200 choices for each experiment involving a choice between A and C_j . In total, we obtained 1000 random choices reflecting preferences between pairs. Based on this, we reconstruct the distribution of distribution γ^{goods} and γ^{income} .

This approach provides only an aggregated measure of γ and requires certain assumptions for its implementation. Firstly, it assumes that errors are random and can be attributed to participant distraction. Since unintentional errors cannot be identified, each choice is assumed to reflect the participant's true preference between the available alternatives. Secondly, it assumes that correct choices outweigh distracting errors. For instance, if we randomly select 200 choices between A and C₄, the fourth choice in the jewelry experiment, and 80% of participants choose C while 20% choose A, the strategy assumes that a majority have a γ greater than 0.75. If all responses were due to distraction errors, the expected split would be 50% for each alternative.

Secondly, we treat each choice within a sequence as independent. This is based on the assumption that errors are based on distraction and not influenced by the order of the choice within a sequence. This assumption is crucial as it allows us to consider choices made within inconsistent sequences. Based on this assumption, we can draw individual choices from the set of all available choices rather than the entire sequence of choices and simulate the empirical frequency of preferences at the aggregate level.

The approach also relies on an additional assumption regarding the distribution of responses within a sequence of five choices. If we continue with the example, we know that 80% of participants have a γ^{jewelry} greater than 0.75, but we do not know how many of them fall within the range of $(1, +\infty]$. Similarly, 20% of responses indicate that γ^{jewelry} is below 0.75, but we require additional structure to

allocate them into the above four ranges.

We make two assumptions in the distribution of responses. First, we allocate responses based on the empirical frequency of preferences obtained from the choice draw (Simulation 1). To illustrate, in the random sample of 200 choices between A and C_4 , assume that 80% choose C_4 and 20% choose A. Furthermore, from the random sample of choices between A and C_5 , we know that 90% choose C_5 and 10% choose A. This distribution of choices based on actual responses suggests that the majority is willing to give up absolute jewelry consumption to increase their relative consumption. We assume this distribution to be true and distribute the 160 that prefer C_4 (80%) into 16 who prefer A over C_5 (*range*(0.75, 1]) and 144 who prefer C_5 over A ((1, + ∞]). We also distribute the remaining 20% choices among the four previous ranges of γ : ($-\infty$, 0], (0, 0.125], (0.125, 0.5], and (0.5, 0.75]. Second, in an alternative (Simulation 2), we allocate responses randomly (50% and 50%). Assuming this allocation, we assign 80 participants to those who prefer A over C_5 and 80 to those who prefer C_5 over A from the 160 that prefer C_4 .

The main results are presented in Figures A7, which simultaneously describe the results from both Simulation 1 and 2 and include the results based on the Baseline Criterion for the set of experiments as a benchmark. Panels (a), (b), and (c) compare the distribution of γ^{income} with the distribution of $\gamma^{\text{insurance}}$, γ^{jewelry} and γ^{car} respectively.

Figure A7 presents the distribution for both alternative simulations. Panels (a), (b), and (c) compare the distribution of γ^{income} with the distribution of $\gamma^{\text{insurance}}$, γ^{jewelry} and γ^{car} respectively. As the distribution is based on the complete sequence of responses, the distributions of $\gamma^{\text{insurance}}$, γ^{jewelry} and γ^{car} are bimodal, and the frequencies are concentrated at the lowest $(-\infty, 0)$ and highest $(1, +\infty)$ ranges of γ . However, compared with the baseline results, γ^{income} presents more uniform distribution. According to this strategy, we confirm that three goods and income are positional. Furthermore, most people are positional – their positionality parameters are positive. Finally, the results reported in the three panels are very strong: changing the relative situation of the individual from the income to the goods noticeably shifts the distribution of the parameter of positionality towards the right.

Furthermore, the mean based on these simulations confirms that these goods are more positional than income. The average γ based on simulation 1 is 63 for Health insurance, 62 for Jewelry, 62 for Cars, and 49 for income. While the average γ based on simulation 2 are 56, 56, 55, and 50, respectively.